Beyond Binary Identification:

Gender Affirming Care and its Utility in Forensic Anthropology

by

Brianne Bertram

A thesis submitted in partial fulfillment of the requirements for the degree of

Master of Arts

Department of Anthropology

University of Alberta

© Brianne Bertram, 2024

Abstract

The role sex plays in forensic anthropology varies from the role gender plays in cultural performance. This thesis looks at how the changes that occur during a transition may be interpreted by forensic anthropologists. Violent crime rates against gender-diverse individuals are high for the small size of the population and variation in gender-affirming can offer potential avenues to assist in forensic investigation. Through a literature review and a series of interviews, this thesis will look at some of the actions transgender individuals may choose to undergo to affirm their gender identity and how these choices may be reflected in the skeletal record. It also recognizes current calls for action by the transgender community and members of the academic community urging forensic anthropologists to deepen their understanding of how individuals who have undergone transition might be recognized postmortem. Working with transgender communities to expand our understanding of surgical intervention in the context of transgender medical care, forensic anthropologists might offer insights about a decedent's history that conventional sex estimation from metric data may not be able to provide.

Preface

This thesis is an original work by Brianne Bertram, supervised by Dr. Sandra Garvie-Lok. The interviews conducted as part of this thesis received research ethics approval from the University of Alberta Research Ethics Board, Project Name "Beyond Binary Identification: New ideas of gender and how they fit into forensic anthropology," No. Pro00133307.

Acknowledgements

I would like to thank my parents for giving me the opportunities that have allowed me to be here writing this thesis and doing research I love. Without your kindness and support, I'm not sure I would have been able to finish. Thank you to my supervisor, Dr. Sandra Garvie-Lok, for taking a chance on me and my forensic interests, I have learned so much in my time with you and I truly appreciate it. I want to thank Tonya Simpson for her help throughout my program and her advice in both writing and careers. I would also like to recognize the work Dr. Fiona McDonald has put into helping me succeed in academia. I would not be in grad school without your assistance, and I am very grateful for your advice and kindness.

Table	of	Con	tents

PREFACE	II
ACKNOWLEDGEMENTS	III
LIST OF FIGURES	VI
LIST OF TABLES	VII
Chapter 1: Introduction	1
Chapter 2: Violence against Transgender Individuals	
Chapter 3: Current Practices in Sex Estimation	
3.1 How do we Define Sex?	
3.2 Current Methods in Sex Estimation	
3.2.1 The Phenice Indicators	
3.2.2 Other Morphological Traits of the Pelvis	
3.2.3 The Morphological Traits of the Skull	
3.2.4 DNA Analysis	
Chapter 4: Thinking about Transgender Skeletons	
4.1 Hormonal Body Modification	
4.1.1 Cross-Sex Hormones, Puberty Blockers and Bone Mass	
4.1.2 Oral Contraception	
4.2 Surgical Interventions	
4.2.1 Case Study – Flaherty, Byrnes and Maddalena (2023)	
4.3 The Archaeology of Castration	
4.4 Conclusion	
Chapter 5: Interview Component: Methods and Results	
5.1 Study Design and Methods	
5.2 Results	
5.2.1 Using Cultural Context and Evidence	
5.2.2 Plastic Surgery	
5.2.3 Concerns about Overreaching	
5.2.4 Considering Gender in Anthropology	
5.2.5 Terminology Use	
5.2.6 Increasing Assessment Confidence	
5.2.7 Collaboration with Transgender and Gender Diverse Communities	
5.2.8 Future Research and Conclusions	
Chapter 6: Conclusion	
6.1 Summary	
6.2 Some Ways Forward	
Bibliography	
Appendices	
Appendix A – Ethics Approval	
Appendix B – Letter of Initial Contact	
Appendix C – Interview Guide	
Appendix D – Informed Consent	

List of Figures

Figure 1 A) an I-shaped osteomy. B) an L-shaped osteomy. (Hann and Kwon 2022)	10
Figure 2 Forehead Masculinization in Ousterhout 2011, p. S11	41
Figure 3 A Sliding Genioplasty presented in Ousterhout 2011, p. S12	42
Figure 4 Patty's 3D-printed skull. (Flaherty, Byrnes, Maddalena 2023)	46

Table 1 Measures advocated by Adams et al. (2023, 154)	9
Table 2 History of Facial Feminization Surgery (FFS) procedures undergone by Patty (Flaherty, Byrnes, and	
Maddalena, 2023)	49

Chapter 1: Introduction.

My interest in sex identification research began when I learned about the small population of individuals born intersex throughout the modern world via the documentary "Intersexion: Boy or Girl" (Lahood 2013). This led me to question my understanding of sex beyond the traditional binary that I was raised with in Western Canada. This pathway continued throughout my studies in anthropology and led me to take courses on the anthropology of gender where I learned about the disproportional and increasing rates of violence affecting transgender individuals. As I developed an interest in osteology, I wanted to learn more about the role hormones play in bone development. From there I wondered what effects individuals who undergo a hormone transition would display in the osteological record and how this would affect forensic anthropologists' ability to identify individuals.

My research aims to determine how the processes involved in gender-affirming care can affect the body and to determine which of those processes are reliable indicators of transition to forensic anthropologists. In order to do this, I first had to become familiar with bone growth and development and the role that hormones play in sexual dimorphism and aging. I also needed to become familiar with the current practices of sex estimation in forensic anthropology and the history behind them. In this process, it was apparent that the current methods were created during a time when gender and sex binaries were not questioned by osteologists; they were assumed. This influenced the way methods developed, looking only for ways to differentiate between male and female without considering the people born outside of this binary. Thinking of the documentary that started my interest in this topic, I looked into the lack of research regarding intersex skeletal development in both forensic identification and clinical research. Unfortunately, the research I did find suggested that distinctive intersex traits would be limited to soft tissue and outside the scope of the thesis. Instead, I focused on how changing the natural hormone cycle during gender-affirming care may result in observable changes to bone.

After considering how sex is defined in current research, I investigated multiple methods of skeletal sex estimation both historically and those used by anthropologists today to better understand how a forensic determination between male and female is made. After this, I identified four areas where a review of the literature can suggest the evidence that transitioning might leave in skeletal remains: pharmaceutical hormonal modifications, facial plastic surgery scarring, surgical implants, and the historic practice of castration to see how the act of emasculation affected skeletal development. As I will discuss below, each of these areas has its own interest for my research question and its own potential strengths and weaknesses as a potential method for identifying the remains of trans individuals.

I conducted interviews to get feedback on the methods I had researched, but also to gain insight on the current practices of professional forensic anthropologists and where they see the future of this research going. Based on the research I had done, I wrote interview questions, an interview guide and completed my ethics application to interview practicing forensic anthropologists. From Fall 2023 to Winter 2024, I conducted my interviews with four participants over Zoom and in person. In the following months, I analyzed their responses for key themes and considered what they suggest about where research in this area is heading.

This thesis presents the results of my work as follows. Chapter 2 discusses the reasons why transgender individuals are an at-risk population in forensic terms and reviews the current calls upon forensic anthropologists to use inclusive terminology in case reports. Chapter 3 contains a discussion of how we can define sex and current osteological practices for determining sex. Chapter 4 considers clinical evidence for the skeletal effects of hormonal

treatment and what the archaeology of castration can tell us about these effects, as well as the impact of surgical modifications, and how all of these may appear in the skeletal record. Chapter 5 is where I discuss my interviews; it contains information about the participants and their professional roles, their recruitment, and the insights my interview research provide. Finally, my conclusion in Chapter 6 summarizes these discussions and provides my overall findings and suggestions for where further research can be conducted.

Chapter 2: Violence against Transgender Individuals

The role of the forensic anthropologist in identifying individuals from their skeletal remains is deeply intertwined with legal systems which have adapted to meet modern ideas of sex and gender, allowing individuals to change their sex on their legal identification. This presents the first new complication in forensic assessments of transgender individuals. The second is the unknown effects transition has on the skeleton. In addition to presenting more complicated cases, transgender individuals are often the victims of violent crimes resulting in murder. The social disenfranchisement faced by transgender individuals makes them vulnerable to individuals seeking to commit violence and isolation leads to the abundance of missing persons cases reported annually.

Doug Meyer shares several personal accounts from LGBT+ individuals who have experienced gender-based violence in his work "Violence Against Queer People: Race, Class, Gender, and the Persistence of Anti-LGBT Discrimination." Multiple individuals reported being cornered and sexually assaulted, accompanied by a verbal assault that focused on their sex or gender along with the implication that they would be interested in the unwanted advance because of said sex/gender (Meyer 2015). Dominique, a twenty-three-year-old transgender woman, describes two incidents of sexual assault, the first of which occurred when she resisted a man's advances and he got aggressive with her, and she feared he would beat her or kill her (Meyer 2015, 80). The second assault occurred when Dominique was engaged in sex work and a man pulled a gun on her and forced her to continue; when she reported to a nearby police car after the event, she was disregarded, and the officers drove in the opposite direction of the assailant's car (Meyer 2015, 81). Eva, a transgender woman, emphasized the victim blaming that transgender women face when she attempted to report to the police. The officers' response included statements that she "couldn't be raped" because she was transgender and the interchangeable use of "he" and "she" to refer to Eva (Meyer 2015, 78). Meyer did not feature any accounts from transgender men, but these ideas are repeated by others throughout his work. While the reasons for LGBT+ discrimination are outside the scope of this thesis, these accounts provide insights into some of the violence LGBT+ individuals are experiencing. While Dominique and Eva are only telling their experiences, stories spread and the fear of not being taken seriously or further abused can make LGBT+ individuals afraid to report assault.

Moving from the personal to the general, a 2015 survey by the National Center for Transgender Equality reported that 46% of transgender respondents had experienced verbal harassment within the past year (James et al. 2016). 9% of respondents reported being physically attacked within the past year and 47% reported sexual assault within their lifetime, 10% of whom reported that this occurred within the past year (James et al. 2016). The American Bureau of Justice Statistics followed up on this report with research on violence resulting from sexual orientation and gender identity and found that the rate of violent victimization of transgender individuals during 2017-2020 was 2.5 times the rate experienced by cisgender individuals, with 51.5 of every 1,000 transgender individuals surveyed reporting themselves as victims (Truman and Morgan 2022). Wirtz et al. (2020) created a comprehensive list of the studies done on gender-based violence in the United States between 2008 and 2016, during which time 165 transgender people were reported murdered in the United States while 2,343 were reported murdered globally. That study found that the estimated prevalence of gender-based violence among trans individuals in the United States reported by individual studies ranged from as low as 7% to as high as 89%; they argue that this large range is partly due to a lack of reporting by victims who do not feel safe or do not want to involve the police (Wirtz et al. 2020). These rates

of violence experienced by transgender individuals are disproportionately high given the number of transgender individuals in the population.

Flores et al. 2021 studied survey data from 2017-2018 looking for patterns of victimization against transgender individuals. The goal of this article was to help raise awareness of this disproportionate violence and call on public health institutions to increase awareness and address the issue (Flores et al. 2021). One pattern they found was that the number of violent incidents reported by transgender individuals was approximately half the actual number of violent incidents experienced (Flores et al. 2021; Gyamerah et al. 2021). One reason for this has been illustrated by the stories of transgender individuals who had negative experiences with certain police officers. In "Violence against Queer People: Race, Class, Gender, and the Persistence of Anti-LGBT Discrimination" we are presented with several interviews of individuals like Eva discussed above, who reported that the gender-based physical or verbal violence they experienced was repeated when they attempted to report these interactions to the police (Meyer 2015).

Distrust between the Queer community and public institutions like the police has a long history. On June 28, 1969, there was a six-day rebellion in New York, N.Y., where the LGBT+ patrons of the Stonewall Inn rebelled against a police raid and anti-homosexual laws (Halkitis 2019). While Stonewall was not the first uprising in response to police actions, it is one of the most recent successful and influential uprisings that had an impact on Western homosexuality laws. Despite the resulting changes to these laws, the tensions between the police and queer community have lingered and stories of individual police officers who enact violence against transgender individuals who attempted to report a crime against them have resulted in a lasting distrust of the police; as a result, many cases go unreported (Meyer 2015).

In 2010 the National Coalition of Anti-Violence Programs found that 50% of the individuals who died of violent crime and were reported to them in 2009 were LGBT+ individuals. Transgender women made up only 11% of the total reported victims but 50% of the 22 hate-motivated murders reported that year (Mitchell-Brody and Ritchie 2010). A global study conducted between 2008 and 2011 estimated that 831 transgender or gender non-conforming individuals were reported murdered (Balzer and Hutta 2012). The murders of transgender individuals occur across the world, illustrating the need to recognize this group and look for ways to assist forensic anthropologists in identifying those who have been victimized.

Also relevant to forensic anthropologists is evidence of repeat abuse. In 2010, the National Intimate Partner and Sexual Violence Survey found that 50% of transgender individuals surveyed had experienced physical partner violence since coming out as transgender (Black et al. 2011). More recently, a review of 85 articles on intimate partner violence found that transgender individuals were 2.2 times as likely to experience physical intimate partner violence and 2.5 times as likely to experience sexual abuse at the hands of their domestic partner than eisgender individuals (Peitzmeier et al. 2020). As noted by the authors, self-reporting does not always provide the most accurate results for this population as approximately half of the individuals approached did not partake in the surveys and thus did not report the violence they have experienced. Taking this into account, the number of transgender individuals who experience physical violence is overrepresented for the small population size and may be even larger than what was reported.

It is important to note that within these statistics is another statistic; approximately half of those who have experienced transgender-related violence are people of colour (MitchellBrody and Ritchie 2010; Gyamerah et al. 2021). The intersection of racial and gender identity is outside

the scope of this thesis, but it needs to be noted that not only are transgender people of colour more disenfranchised, but they are particularly targeted when they appear obviously transgender or do not conform to the predominant gender ideologies of the dominant culture (Jefferson, Neilands, and Sevelius 2013). Using data initially collected for a longitudinal study on HIV in trans women, Gyamerah et al. (2021) used the additional data collected to look at the hate crime experiences of these women between 2016-2018. They found that again, only half of these women (51.1%) reported their experiences to the police, but their data set also took ancestry into account and found that of those who reported experiencing violence, Black and Latina trans women were more frequently victims of violence with a weapon and sexual assault (Gyamerah et al. 2021). This thesis will focus on the skeletal remains of transgender individuals, but the ability to access resources and the intersection between race and sex are crucial to a broader understanding of profiling.

The rates of violence against this group were a driving factor for my thesis research. There are a handful of new works calling specifically for research in this area. One that became a major influence on my interview research is a 2021 publication resulting from a master's research project reporting that 82.1% of 115 forensic anthropologists surveyed had discussed the difficulties of emerging gender identities, the socially changing relationship between sex and gender and how these relate to forensic anthropology with their students (Tallman, Kincer, and Plemons 2022). Another 60% of respondents reported discussing potential issues of identifying transgender decedents (Tallman, Kincer, and Plemons 2022). This report will be discussed in more depth in Chapter 5, as the survey research inspired some of my interview questions and I wanted to get more detailed answers on these topics than a survey could provide. Adams et al. (2023) reviewed current literature on the anti-trans violence "epidemic" and called upon forensic anthropologists to recognize the role they play in "cisgenderist practices and systems: in casework, research and education" (142). Adams and colleagues argued that the work forensic anthropologists do is rooted in cisgenderism that is ingrained in the academic, health and legal sectors. They call on forensic anthropologists to act in four ways: by being transparent about the limitations of forensic anthropological analysis, by mitigating biases and misinterpretation of skeletal analysis, by making use of a bio-cultural profile, and by actively combating misgendering within case files and investigations (Adams et al. 2023). The limitations of skeletal analysis will be discussed in the next chapter where I look at how sex is defined within forensic anthropology. Adams et al. (2023) applied Evans and Kneppers' (2020) to create their suggestions for best practice.

Table 1 Measures advocated by Adams et al. (2023, 154)

Tiers of the Diversity and Inclusion Model (Following Evans and Knepper [129])	Application to the Forensic Anthropology Classroom	
Tier 1: Representation of gender	 Interject gender diversity readings in course syllabi and lectures. Include guest speakers from the TGD community on the course schedule. Meaningfully include specific examples showcasing how gender diversity is represented in forensic cases. Use "they/them" pronouns when referring to unidentified decedents and persons of unknown gender identity in lecture materials. 	
Tier 2: Gender-sensitive course environment	 Create a welcoming classroom environment by outlining expectations for inclusivity and discussion. Assign readings from diverse authors that emphasize multiple perspectives. 	
Tier 3: Embedded application	 Challenge cisnormativity in readings and discussion. Encourage sharing of experiential knowledge. Promote self-reflection for both students and instructor. Commit to application of queer theory in course design and instruction. Design assignments that challenge binary conceptions of sex and gender. 	
Tier 4: Gender acculturation	 Acknowledge that language, visuals, terms, and phrases are often gender-coded and re-examine usage in the classroom and syllabi. Avoid using images, quotes, and phrases that reinforce gendered stereotypes. Critically examine all examples and images used in coursework to ensure that underlying messages are inclusive. 	

In 2024, Adams and eleven other forensic anthropologists again called upon the field of forensic anthropology to engage in more activism and advocacy for transgender communities

(Adams et al. 2024). One goal of this publication was clarity on the activism they had previously called for, explaining that activism does not need to mean protesting and marches, but rather an individual choice to engage in measures like the shift in terminology from simply 'sex' to 'assigned sex at birth' or 'osteological sex' in reports (Adams et al. 2024). This article also covers other disenfranchised groups, as well as issues of intersectionality, primarily advocating that awareness is the first step towards activism (Adams et al. 2024). For those more interested in activism, Adams and colleagues suggest forensic anthropologists look at the history of forensic anthropology pedagogy and evaluate the role those prominent ideologies of sex and gender had in its development, comparing this to the re-evaluation of ancestry within the field. The history of the development of sex estimation methods will be further looked at in Chapter 3.

In Chapter 4, I discuss plastic surgery and the visible traces it may leave in the skeletal record for forensic anthropologists to interpret. On this subject, Flaherty, Byrnes, and Maddalena (2023) were presented with an opportunity to analyze the CT scan of a transgender woman who had undergone 13 procedures to feminize her face including three rhinoplasties, cheekbone reduction and reconstruction, tracheal shave, gonial angle reductions, and orbital rim shaving. The transgender woman, called Patty in their article for confidentiality, approached Flaherty and team and asked them to analyze her CT scan using the program FORDISC to determine if the new measurements of her skull after these surgeries would present her as male or female. The CT scan, shown in Figure 4 of this thesis (page 48), was used to generate twenty-seven measurements that were inputted into FORDISC and resulted in a classification of White Male with a validation of 92.2% (Flaherty, Byrnes, and Maddalena 2023). While this experiment did not yield the results Patty hoped for, it provides information about our potential ability to analyze the evidence of surgical modification on the skull which I will discuss more in Chapter 4. One

important example is that forensic anthropologists have been approached by a civilian who was concerned about her ability to be correctly identified if something happened to her. Patty's concern is an example of the individual fear felt by transgender individuals and their desire to engage in more community-led research.

The disenfranchisement of transgender individuals is not a new concern. The violence against this group is well documented, even despite reports that suggest only half of the actual incidents are reported. Forensic anthropologists are not unaware of this group, and the desire to find ways to identify them and otherwise support their needs is expressed by a handful of academic articles calling for a change in terminology (Flaherty et al. 2023; Adams et al. 2023; 2024). The issue has also been raised by a transgender individual who approached scholars, looking for ways to help identify her community members (Flaherty, Byrnes, and Maddalena 2023). The concern has been voiced and now it is up to forensic anthropologists to answer the call, starting by looking at the history of sex estimation in our discipline and how we define sex.

Chapter 3: Current Practices in Sex Estimation

The conversation about sex is much larger than can be covered by the scope of forensic anthropology alone due to our narrow focus on the skeleton. Clinical definitions of sex agree that it is determined by several interplaying factors, but in forensic anthropology, we look to the sexual dimorphism that occurs in the human skeleton during life. Many traditional methods have forensic anthropologists score skeletal morphology on a 5-point scale of masculine to feminine working on the assumption that human sex is binary but recognizing that skeletal morphology is not, with individuals who fall between the two extremes described as male, possibly male, unknown, possibly female, or female (Buikstra and Ubelaker 1994; Blau 2017; Klales 2020). This scale presents the first issue forensic anthropologists face: how do we define what sex is in order to score it?

3.1 How do we Define Sex?

The generally accepted definition of sex is that it is biologically determined and is different from gender, which is a social construct. This idea was first popularized by Sigmund Freud in the 1895 "Studies on Hysteria," where he expressed the opinion that sex was something completely separate from gender (Bourseul 2014; Alfandary 2019). Freud argued that gender was a social construct that shaped the way people were supposed to behave, whereas sex referred to the physical transformations of the body and the cognitive developments that occurred alongside them (Freud 1895; Bourseul 2014; Alfandary 2019). This notion has continued in modern feminist scholarship. An example of this thought is Dr. Judith Butler's "Bodies that Matter" where she addresses the issue that the category of sex is almost always considered equal to gender although that is not the case (Butler 2011; Reid 2018). Butler demonstrates the connection between the physical body and gender expression and how these tie into specific

cultural constructions of gendered roles (Reid 2018). The relationship to culture is one of the fundamental differences between sex and gender, the other being that sex refers to a set of biological characteristics. Despite this difference, sex and gender are often used interchangeably in casual discourse and every culture has its own prescribed opinions on how individuals should express their gender through actions and appearance in socially acceptable ways.

Biological sex can be defined by a combination of biological factors, including phenotypic expression, chromosomes, and gamete production (Goymann, Brumm, and Kappeler 2023). Due to the complexity of biological sex, many argue that one simple defining factor is not enough (Garofalo and Garvin 2020; Klales 2020). Rather, biological sex is made up of the interplay of these factors; for example, an individual may be born with externally malepresenting genitalia but possess ovaries, a female reproductive organ.

Beside male and female, the third sex category largely accepted in western academia is intersex, coined by Richard Goldschmidt in 1917. Looking at the larger animal kingdom, Goldschmidt framed his argument for the term intersex by acknowledging the variety of sexual dimorphism that can occur in various species and the ability of some species to undergo sex changes in response to environmental stimuli. He concluded that sex characteristics are not decided until the point of fertilization, especially in the case of some insects which do not require two individuals for reproduction (Goldschmidt 1917). From there, the term intersex was derived to mean in-between sexes, chosen for its ability to be translated into other languages but also because Goldschmidt found that humans he characterized as intersex presented with a wide variety of organ and hormone variation between individuals making it challenging to describe the specifics of the condition. This term was not used by clinicians until the later part of the 20th century and was used interchangeably with hermaphrodite, rather it was the adoption of the term

by the intersex community in the 1990s that led to the medical community prioritizing the term (Feder and Karkazis 2008).

Some scholars argue that the multifaceted definition of sex and intersex individuals means the sex binary should be questioned. One scholar's argument against a sex binary is presented in "The Five Sexes," by Dr. Anne Fausto-Sterling (1993), where she argues that two biological sexes are not enough to accurately represent the full range of human variation, beginning what continues to be a discussion 30 years later. She defines sex on three levels, cellular (sex-specific cells, or X and Y chromosomes), at the hormonal level and at the anatomical level; she goes on to explain that each of these can be misaligned with gender identity. The original article proposed three additional categories: true hermaphrodites, male hermaphrodites and female hermaphrodites (Fausto-Sterling 1993). The term hermaphrodite comes from the name of two Greek Gods, Hermes and Aphrodite, who were said to have parented a child who was born male and fused with his true love resulting in a body with the organs of both sexes (Fausto-Sterling 1993). Later, in "The Five Sexes, Revisited," FaustoSterling (2000) called for a switch from 'hermaphrodite' to 'intersex.' It was hoped that this shift would eliminate the idea that these expressions of gender are deviations from a natural binary, encouraging the understanding that intersex in some form has always existed and should therefore be considered its own category, not a sub-category of binary sex (Fausto-Sterling 2000).

The frequency at which we see the combination of phenotypic traits and gamete production resulting in what we describe as male or female has meant that the binary sex system has remained dominant in clinical practice despite Fausto-Sterling's (2000) estimation that 1.7% of children are born outside of this binary and can be considered intersex. Other scholars argue that due to the nature of human reproduction, the process of anisogamy, the ability of sexually reproducing organisms to produce either large gametes (eggs) or smaller mobile gametes (sperm), only allows for two sexes (Goymann, Brumm, and Kappeler 2023). They continue by highlighting sex roles in other species which may vary, but note that the gamete production remains consistent and argue that these traits do not reflect male or femaleness (Goymann, Brumm, and Kappeler 2023). Using this definition on the example of an individual born with male genitalia but large gamete-producing organs, we run into the issue that this is not sufficient to describe natural variation (Rehmann-Sutter et al. 2023). Rehmann et al. (2023) also bring up the issue of those born infertile who are still able to be categorized as male or female due to sex being more complex than one definition allows for.

An example of the critical responses to Fausto-Sterling's work is Sax (2002), who approaches Fausto-Sterling's work from the angle of the chromosomal sexes. There are largely considered to be two chromosomal sexes - XY and XX, male and female respectively - which are the most common chromosome presentations and have contributed to the current binary sex model. However, the definition of intersex used by Fausto-Sterling to expand into five sexes is considered by Sax (2002) to be inaccurate because it considers those born with Klinefelter syndrome, Turner syndrome, and XXY syndrome as intersex. If this definition is changed, as Sax proposes, to mean only that either chromosomal sex and phenotypic expression do not align or that phenotypic expression is not able to be classified as either male or female, the number of predicted intersex individuals drops to 0.018% (Sax 2002). Using Fausto-Sterling's definition, these chromosomal syndromes would be defined as three additional chromosomal sexes, which have a range of mental, physical, and hormonal effects on an individual. To evaluate this argument, we need to understand each of these syndromes and their presentations. The following descriptions of these syndromes focus on the potential skeletal impacts that may be visible.

X0 syndrome describes the situation when one of the two X chromosomes is fully or partially missing; it was named Turner syndrome in 1938 by Dr. Henry Turner (Roberts and Fechner 2020). Turner syndrome presents osteological symptoms as a result of its associated SHOX haploinsufficiency. The SHOX protein plays a key role in bone growth and maturation, so its low level in X0 individuals affects the skeleton during development (Marchini, Ogata, and Rappold 2016). Turner syndrome individuals can present with shortened 4th and 5th metacarpals, altered body proportions with a high upper-to-lower segment ratio (also described as sitting height), and dorsal subluxation of the distal ulna related to epiphyseal growth arrest of the distal radius, also known as a Madelung deformity (Atik, Çoğulu, and Özkınay 2016). Treatment for Turner syndrome includes the use of growth hormone therapy, but the long-term outcomes of this treatment are still unclear (Jodarski et al. 2023). Turner syndrome also puts individuals at higher risks of developing cancer and autoimmune conditions, which may present as additional lesions on the skeleton; these lesions have been studied in the field of osteoimmunology (Jodarski et al. 2023).

Klinefelter syndrome refers to the situation in which an XY individual is born with at least one additional X chromosome. This may present as either 47,XXY syndrome, 48,XXXY syndrome, 48,XXYY syndrome, or 49,XXXXY syndrome, but all are grouped as chromosomal variants of Klinefelter syndrome. Klinefelter syndrome occurs in approximately one in 1,000 XY individuals born; almost all of these individuals will be infertile, but phenotypic effects are far more varied (Wattendorf and Muenke 2005). Klinefelter syndrome is primarily characterized by hypergonadotropic hypogonadism (abnormally low hormone production in the gonads) and neurodevelopmental deficits, but tall stature is also considered a classic feature of Klinefelter syndrome in individuals with the 47,XXY karyotype (Wattendorf and Muenke 2005). However, if the 49,XXXY karyotype is present, it can result in short stature as well as other skeletal traits: cleft palate or a flattened nasal bridge (Rajabzadeh, Taheri, and Jazayeri 2022). Additionally, the even rarer 48,XXXY karyotype has presented with mild craniofacial dysmorphism, taurodontism (apically displaced teeth with decreased root length, less visible cervical constriction, and a larger than average pulp cavity), clinodactyly and radioulnar synostosis or elbow deformity (Joseph 2008; Atik, Çoğulu, and Özkınay 2016).

XYY syndrome, named for the presentation of an additional Y chromosome in XY individuals and sometimes referred to as Jacob's syndrome, is characterized by neurodevelopmental phenotypes including developmental delay, increased mental health risk, and cognitive impairment (Jodarski et al. 2023). The phenotype of tall stature may accompany XYY syndrome, but overall physical characteristics are considered less evident than the other two syndromes discussed and have a large variation among individuals that has resulted in the underdiagnosis of the syndrome historically (Jodarski et al. 2023).

While these three chromosomal presentations can have skeletal markers, these are seen in only a small fraction of the already small population of individuals, one best described by the current description of the chromosomal fluctuations as part of larger syndromes. This brings into question whether any additional clarity would be added to forensic anthropology by introducing Fausto-Sterling's proposed additional chromosomal sexes and veering away from a sex binary. Due to the infrequency of these chromosomal syndromes, the many problems that accompany them and the range of human natural variation within them, I believe this would only increase confusion in the assessment of individual sex. Rather, any presentation of these syndromes would be better noted in terms of pathologies and individualizing features. The presence of accompanying pathologies is better considered as an individualizing feature that will provide examiners with additional information about the decedent. Additionally, many of these syndromes are described by doctors in terms of their relationship to the traditional sex binary. For example, Klinefelter's is described as a syndrome which occurs in 1/1000 boys; this suggests that the development of the individual's secondary sex characteristics will still fit within the recognized sex binary, again making noting evidence of these syndromes more useful as identifying features because the individual would have been labelled as a boy in life.

3.2 Current Methods in Sex Estimation

Sex estimation methods used in biological anthropology focus on classic male and female phenotypes and then score skeletal traits on a continuum between them. This is in part because, during the creation of our standards of practice, the two sexes male and female were relatively unquestioned. While the conversations of Freud and others about the mental aspects of sex and gender were being discussed, the biology of the two sexes remained unquestioned. In the field of anthropology, our methods look at sex in the same way: there is an understanding that sex and gender are two separate yet often related categories, but sex is an unquestioned binary. The most commonly used methods for sex estimation primarily use the innominate and the skull to complete their assessments as these are the most consistently sexually dimorphic parts of the skeleton (Yang et al. 2020; Klales 2020). Even the term dimorphic reflects the dominant assumption in the field that sex is either male or female.

For both sexes, practitioners can use methods that rank each skeletal feature along a comparative scale to assess how masculine or feminine it is. We can also use quantitative measures like 3D modelling and metric analysis to determine the sex of an individual. It is

important to add that no method is perfect; even the best do not offer 100% accuracy and as a result, the best practice is to use multiple indicators to generate a composite estimate with the decedent assessed as male, female, likely male or female, or undetermined. I will review the specifics of scoring some commonly used morphological traits and how these are considered to range from masculine to feminine.

3.2.1 The Phenice Indicators

Many biological anthropologists rely on the longstanding practice of applying the Phenice traits to estimate the biological sex of skeletal remains. The Phenice traits are three morphological distinctions in the pelvis described by T.W. Phenice in 1969. Phenice created his approach using these traits to help anthropologists working with fragmented or delicate ancient remains, but the practice has since been used for modern contexts as well. His methods were based on observation, looking for the presence or absence of traits associated with either males or females. However, as he noted some of these traits can be possessed by both sexes but have different expressions. These methods are an important foundation of biological anthropologists' ability to conduct sex estimation of the pelvis and were included in the 1994 *Standards* volume published by Buikstra and Ubelaker, becoming the standard across the field.

The Phenice traits are the ventral arc, subpubic concavity, and ischiopubic ramus ridge. The ventral arc is located on the ventral surface of the pubic bone. It is an elevated ridge along the surface that is prominent in females and slight in males (Buikstra and Ubelaker 1994). In the 1960s, when the ventral arc was originally scored by Phenice, it was only observed in known female innominates. Phenice did record instances where males had a similar ridge, but he argued that when properly observed, the male ridge is either extended from the pubic crest to the pubic symphysis or extends over the pubic symphysis inferiorly and ends superior of the subpubic angle, forming a similar angle to the ventral arc (Phenice 1969, 298-300). Since then, when scoring the ventral arc, a complete absence is considered to indicate a male pelvis, but methods have since developed that allow ventral ridge prominence to be scored to further determine if the individual is female or unknown (Moore 2013; Ubelaker 2018; Klales 2020). The ventral arc is considered to have the highest accuracy of the Phenice traits, over 85%, due to the relative lack of observer error (Kilmer and Garvin 2020).

Phenice first used the subpubic concavity, located on the dorsal side of the subpubic region, to determine sex. He looked at the border of the ischiopubic ramus to determine the level of concavity or convexity present. Initially, it was determined that if the border is concave the individual is female, and if it is convex it is male (Kilmer and Garvin 2020). As with the previous Phenice trait, the male-female distinction is not quite as simple as it initially seemed due to the natural range of individual variation caused by different directions of growth occurring at the middle of the ischiopubic ramus (Buikstra and Ubelaker 1994). The medial border of the ischiopubic ramus is its own trait scored by Phenice, described as narrow and crest-like in females, whereas male ridges are considered broad and flat (Buikstra and Ubelaker 1994). Phenice (1969) believed that the ischiopubic ramus was the least accurate of the three methods, and today the ischiopubic ramus is considered to be a supportive classification with a 75.8% success rate when scored alone (Klales, Ousley, and Vollner 2012, 111).

The Phenice traits have been tested numerous times over the years for their validity and repeatability. Klales et al. (2012) found their 75.8% success rate by taking the average of four students: two graduate students with experience in sex determination, one student who was familiar with an undergraduate course, and the last student with no previous knowledge of estimation or osteology. Later in a review of studies testing the Phenice methods, Klales et al.

(2020) found that the traits average between 80-99% effective, though one outlier study reported only 59-83% accuracy. The Phenice traits have been repeated countless times and their continued accuracy is why they remain a prominent method in today's standards of human sex identification.

3.2.2 Other Morphological Traits of the Pelvis

The greater sciatic notch and the preauricular sulcus are the two other pelvic traits scored in the 1994 *Standards* volume published by Buikstra and Ubelaker, which has become a widely used standard for data collection from skeletal remains (Klales 2020). The addition of these two measurements reflects their pattern of sex-specific variation and when used alongside the Phenice methods is reported to achieve an accuracy rate of 95% (Moore 2013).

The greater sciatic notch matures to be deep and narrow in males, creating a "J" or "U" shape, and is wide and shallow in females (Houghton 1974). The greater sciatic notch is assessed on a 5-point scale, where 1 is feminine and 5 is masculine (Buikstra and Ubelaker 1994). The greater sciatic notch is praised for its ability to survive intact making it a popular trait to score in archaeological contexts (Houghton 1974). It has been noted that in older females, the greater sciatic notch can come to take on masculine traits and should be used in tandem with other methods when possible to conduct sex estimation (DesMarais, Obertova, and Franklin 2024). However, it has since been argued that changes to the greater sciatic notch are due more to extrinsic factors and population affinity than age itself (DesMarais, Obertova, and Franklin 2024).

The preauricular sulcus runs along the interior edge of the auricular surface and has been argued to have two forms, one of which only occurs in females (Houghton 1974; Buikstra and

Ubelaker 1994). The first is described as giving "the impression of being formed by the coalescence of a series of pits, or small craters, in the bone...the floor of the [sulcus] is uneven... with a undulating margin" (Houghton 1974, 381). The second form of the sulcus is more variable and is described commonly as short, narrow and straight-edged though it can be wider, when it is it tends to have a rough lateral edge and is only found in females who have had at least one child (Houghton 1974; Karsten 2018). The preauricular sulcus is scored on a 0-4 scale, with the absence of a preauricular surface as 0 and the numbers 1 to 4 assigned to sulci of different appearances. For example, 4 describes a narrow preauricular surface that is less than 0.5 cm and a "shallow, and smoothed-walled depression" (Buikstra and Ubelaker 1994). In a study of 500 left os coxae of known sex, it was found that 75.8% of individuals had a correct assessment based on the preauricular sulcus (Buikstra and Ubelaker 1994). The same study found accuracy for 62.84% of males and 89.96% of females being assessed and females were found to have a wider, longer sulcus than their male counterparts (Buikstra and Ubelaker 1994). The preauricular sulcus is an example of a feature where scoring traits on a scale is more informative than only acknowledging the presence or absence of a trait.

These traits were devised to determine male or female from the pelvis but recognize the large range of natural variation that can present; this is why traits are often scored on a scale as opposed to strictly present or absent. The robusticity of the pelvis allows it to preserve well and the sexual dimorphism it presents makes it one of the most useful skeletal regions for sex determination. The methods discussed above have been tested and repeated and found to be consistently accurate even when performed by those with minimal background in sex identification. In developing methods of pelvic estimation, the morphological traits were devised

to prevent user error and are best used in tandem with the other traits listed to identify sex with as many factors as possible to allow for natural human variation.

3.2.3 The Morphological Traits of the Skull

The other sexually dimorphic component often used in the sex estimation of a skeleton is the skull. Once again, Buikstra and Ubelaker (1994) have set the standards in modern best practice with their five-point scale measuring the nuchal crest, mastoid process, supraorbital margin, glabella, and mental eminence. The skull is also often chosen to assess ancestry due to the presentation of population-specific features; these features may cause challenges in sex estimation (Klales et al. 2020). For example, if all of the population has inflated mastoid processes and large supraorbital margins, individuals may be scored as male if considered alone but female when considered within the context of the population.

The nuchal crest is located on the posterior aspect of the cranium, where the attachment area of the nuchal musculature is located (Gill, Jeffreys, and Werrett 1985; Bukyya et al. 2021). It is scored from minimal expression to a large nuchal crest with considerable projection from the bone; a marked expression also features a "hook" or "well-defined bony ledge" (Thomas 2020). The mastoid process is located below the external auditory meatus. It scores on a 5-point scale where 1 is a minimal expression, female, and describes a process that only projects a short distance below the external auditory meatus (Latham and Miller 2018; Thomas 2020; Turingan et al. 2020). A 5 is a large mastoid process that appears substantially larger than the external auditory meatus and is considered typically masculine (Ubelaker 2018; Thomas 2020; Latham and Miller 2018). The supraorbital margin is located on the frontal bone and defines the superior border of the eye socket; minimal expression scores as a 1 and features a margin that can be described as sharp when you run your finger along it. This varies from the extreme masculine

expression scored as 5, where the orbital border is rounded. The glabella is located on the midline of the frontal bone near its inferior border, between the eyes. Often it is also called the brow ridge. A score of 1 is given to a minimally pronounced brow ridge that borders on being considered smooth as it has little to no projection. A score of 5 describes the maximum expression, a large protrusion that is rounded, and sometimes described as loaf-shaped (Ubelaker 2018). The last standard trait is the mental eminence located on the anterior midline of the mandible. It is scored from minimal expression with little to no protrusion of the eminence to a massive protrusion said to occupy the majority of the anterior mandible (Buikstra and Ubelaker 1994). These traits are best used in tandem with other methods, including estimations based on the pelvis and the use of DNA to support findings when possible.

Similar to the pelvic traits, these cranial traits have been tested multiple times across the world but findings have been more varied than the pelvis due to the varying sexual dimorphism present across populations (Rogers et al. 2024). In a study of 34 skeletons of south Asian ancestry, the traits scored a 63% accuracy rate for correctly classifying males and females. Interestingly, this study found 100% success rate in correctly identifying female remains, but only 35% for males (Rogers et al. 2024). In a study of 192 known skulls from the Ottoman period, accurate classification rates ranged from 72-86%, again favouring females as more accurate than male assessments (Yaşar and Sağır 2023). When determining sex, anthropologists must also consider ancestry and compare their findings to previous research on populationspecific morphology to determine if the population was more gracile or robust; without this consideration, there can be confusion and bias towards one sex or the other (Kelley and Tallman, 2022).

3.2.4 DNA Analysis

To support their morphological findings, anthropologists can use DNA analysis to determine the sex chromosomes present in an individual. In 1984 Alec Jeffrey was the first to use DNA analysis for identification in the field of forensic sciences (Gill, Jeffreys, and Werrett 1985; Bukyya et al. 2021). This had major implications for the field of forensics as it allowed for the sex estimation of remains that lack the morphological features described above, like juvenile remains or fragmentary remains. Prior to this discovery, forensic anthropologists and our morphological techniques were the primary sources of information for sexing human skeletal remains.

DNA analysis looks for markers of the Y chromosome only present in men (Skoglund et al. 2013; Dash, Rawat, and Das 2020). One method applied globally uses the amelogenin gene, responsible for the creation of the tooth enamel matrix, to look for the AMELY and AMELX homologues related to men and women respectively (Dash, Rawat, and Das 2020; Gamble et al. 2024). The analysis looks for sex-specific sequences in the gene; if the Y specific fragment is detected, the individual is determined to be male (Dash, Rawat, and Das 2020). This method is not flawless as the absence of the AMELY marker due to poor preservation can result in misidentification of males as female; the method can similarly misidentify men with deletion or mutations in the AMELY region. To combat this, an alternative method based on amelogenin was tested by Gamble et al. using the protein rather than the DNA sequence (2024). They suggest that this method may solve the issue of false negatives caused by deletion, and is a relatively inexpensive and highly accurate method for sex determination (Gamble et al. 2024). Another solution for false negatives is alternative markers for sex determination including Steroid sulfatase (STS), the SRY gene, TSPY gene family, DYZ1 or Alu, all of which involve distinctive sequences on the Y chromosome and use these for sex determination (Dash, Rawat, and Das 2020). Anthropologists argue that whenever possible, DNA sampling should be used in tandem with other methods of sex estimation for the best result (Ubelaker 2018; Thomas 2020; Latham and Miller 2018).

Methods of sex estimation using DNA may be unable to detect the presence of the presence of Klinefelter, Turner, or XXY syndrome causing deviations in data (Skoglund et al. 2013). However, Roca-Rada et al. (2022) used DNA sequencing to look specifically for the presence of X, Y and autosomal chromosomes in archaeological individuals with the goal of mapping them. They found one individual with evidence for a 47,XXY genotype and after examining the remains for the typical morphology described in detail in Section 3.1, they determined that this individual had Klinefelter's (Roca-Rada 2022). Moilanen et al. (2022) used a similar method on a Finnish archaeological individual uncovered in 1968 whose sex has been debated since its discovery. They created their own approach using down-sequencing methods to determine if the DNA sequences recovered best matched XX, XY, XXY, or a contaminated sample (Moilanen et al. 2021). They found overwhelming evidence of genetic patterning which resembles the modern XXY karyotype, again an individual with Klinefelter syndrome (Moilanen et al. 2021). While genetic syndromes are not tested for upon initial examination of sex in DNA, the work done by archaeologists provides a potential solution to the problems raised in Section 3.1, the wide range of sex presentation and physical attributes individuals may naturally present with.

The methods currently used by forensic anthropologists recognize the natural human variation that occurs but work to fit presentations of this variation into a binary system. The standards of sex estimation used are excellent at determining the average individual, but they do not take into account the new ways individuals can modify themselves to express sex in addition to gender or the unique ways individuals who are born intersex can present. It is stressed throughout the publications cited in this chapter that whenever possible all three of the above methods (pelvic morphology, cranial morphology and DNA) should be used together for analysis of modern remains to offset any unreliability from one method alone. Through the use of morphological methods and DNA, anthropologists can accurately determine the sex of individuals who fit within the traditional sex binary, but what about those who have taken active steps to disrupt or alter this process as is the case with some transgender individuals?

Chapter 4: Thinking about Transgender Skeletons

This chapter covers a broad scope of knowledge, looking for ways to observe evidence of transition in the skeletal record. First looking at bone mass density and puberty blockers, I evaluate what I call hormonal body modification, the intentional modification of the body's natural hormone production process. This may be in the form of a disruption in hormone production, like a puberty blocker or an event like castration, or through supplementation using hormone treatments. Hormonal body modifications may be accompanied by a physical modification, like gender-affirming surgery, another source of changes that the forensic anthropologist might be able to see on examination. This chapter will evaluate these potential sources of skeletal difference for their visibility and usefulness to the examining forensic anthropologist.

One method mentioned throughout the chapter is the use of dual x-ray absorptiometry (DXA) to measure bone mass density (BMD). DXA measurements are reported as bone mineral content (BMC, g) or areal bone mineral density (aBMD, g/cm²) which can then be compared to reference values from healthy individuals of similar age, sex, and ethnicity (Bachrach and Gordon 2016; Ramchand et al. 2019). A z-score is calculated using this age and populationspecific reference data. The factors helping to determine an individual's score are complex as

BMD is affected by the individual's diet, exercise, and hormone production (Hale and Ross 2018). For those who are older, a T-score is evaluated to consider risks of osteoporosis (Bacharach and Gordon 2016). Sometimes, more than this score is required to assess bone density. For example, if a juvenile being measured experienced chronic illness over childhood resulting in delayed pubertal development, this individual cannot be directly compared to others

their age as the lack of sex steroid exposure during pubertal development will have impacted their BMD (Bachrach and Gordon 2016). While there is currently no single agreed-upon method to evaluate BMD in situations of restricted growth, using z-score measurements to compare an individual to the height of peers in the same category will help to identify the growth-restricted children (Specker and Schoenau 2005; Bachrach and Gordan 2016).

Hormonal treatment for gender-affirming care relies on a category of treatments broadly described as Cross-Sex Hormone (CSH) therapy. Sometimes also referred to in the literature as Gender-Affirming Hormone Treatment (GAHT), this is the pharmaceutical administration of hormones to help an individual achieve a physical appearance that matches their gender identity and expression (Yun, Kim, and Lee 2021; D'hoore and T'Sjoen 2022). Treatment using gender-affirming hormones, like Gonadotropin-Releasing Hormone analogues (GnRHa) or testosterone, provides adolescents with a relatively reversible method of altering their body to achieve their desired appearance (Nos et al. 2022). Many of these pharmaceuticals have additional uses; for example, GnRHa is also used to treat endometriosis in menstruating women (Surrey 2023). Hormone treatments are continued in different formulations and smaller doses throughout life to maintain results.

Another term used throughout this chapter is gender-affirming surgery; this is when an individual who is experiencing gender dysphoria or discomfort reaches out to a medical professional to alter their physical features to closer resemble the image of themselves they desire (Agochukwu-Mmonu et al. 2022). This may encompass a range of surgical procedures including facial feminization and facial masculinization surgeries, a range of plastic surgery procedures on the face to feminize or masculine appearance (Agochukwu-Mmonu et al. 2022). In section 4.2, I will look at these procedures and evaluate how they may impact the skull in
ways a forensic anthropologist could observe. The other surgical intervention evaluated in the section is the insertion of implants for gender affirmation, including penile implants and breast implants. Most surgical implants have an assigned and registered serial number which would assist in identification.

The last section of this chapter looks at how a dramatic hormone change may present in the skeletal record by looking at the effects of castration on archaeological skeletons. While castration was not a gender-affirming procedure in these past societies, rather being associated with a social role or profession, it still caused a dramatic halt in testosterone production that has a visible effect on the skeletons examined today. Looking at these effects, I will compare the archaeological record to what forensic anthropologists might see when observing a decedent who had undergone a radical decrease in testosterone production during their life.

This chapter evaluates various methods of modification for their visibility in the skeletal record. To do this I will outline the existing literature on each topic and what is known about osteological presentation and discuss potential usefulness to forensic anthropologists.

4.1 Hormonal Body Modification

Hormonal body modification, particularly pharmaceutical interventions including puberty blockers and oral contraception, can affect the density of an individual's bones. The effects of many pharmaceuticals on BMD are very well studied because of concerns over preventing osteoporosis caused by premature drug-triggered bone loss. For patients on some medications, regular DXA scans are recommended to monitor BMD changes (Sheu and Diamond 2016). The effects of medication-induced osteoporosis are well outlined in Panday, Gona, and Humphrey (2014) and Alahmari et al. (2023). Low BMD can serve as a predictor of increased fractures in all ages but is most well-researched in older adults with osteoporosis. Osteoporosis is a common

chronic bone disease characterized by the deterioration of bone, including reduced bone mass; this weakens the bone and causes an increased risk of fractures (Sözen, Özışık, and Çalık Başaran 2016; Buikstra 2019). However, osteoporosis cannot be diagnosed from BMD data alone; diagnosis of osteoporosis requires evidence of secondary fractures to be considered (Bachrach and Gordon 2016; Buikstra 2019). Despite this challenge, research on osteoporosis provides an excellent base of understanding for the appearance of decreased bone mass density. Hale and Ross (2018) present three forensic case studies where DXA scans assisted forensic anthropologists in their analysis; they argue that DXA provides additional insight into fracture patterns, history of starvation or neglect, and other aspects of a decedent's lived experience.

Puberty blockers, cross-sex hormones and oral contraceptives are pharmaceuticals that are used by individuals for gender affirmation and have known effects on BMD (Ferguson 2020; Twohey and Jewett 2022). The next sections will outline how these drugs may affect bone mass.

4.1.1 Cross-Sex Hormones, Puberty Blockers and Bone Mass

Most of an individual's bone mass is acquired during the first 18 years of life and rates of bone mass acquisition increase during puberty when the actions of growth hormones and sex steroids in the skeleton are at a high (Almstedt, Shoepe & Snow 2005; Ciancia, Dubois, and Cools 2022). A key hormone that sees increased production during puberty and has a direct effect on bone mass is estrogen. BMD reaches a peak in the lumbar spine of young women around age 16, continuing to increase at some other sites until approximately age 25 when it begins to decrease throughout the rest of their lives (Almstedt, Shoepe & Snow 2005). Wang et al. (2006) measured serum levels of testosterone and E2 (estradiol, the key form of estrogen), in 258 healthy girls aged 10-13 and compared these to tibial shaft CT scans. They found that E2 and testosterone had different relationships to bone mineral density (Wang et al. 2006). E2 inhibits

bone resorption during rapid growth periods like puberty, as well as after menopause when it is present at high concentrations (Riggs, Khosla, and Melton III 1998; Wang et al. 2006). While this is a single study, focused on young girls around menarche, it is an example of the importance of hormones released during puberty for determining bone morphology and BMD.

Since medically monitored transitions began, doctors have used cross-sex hormones to supplement the loss of natural sex steroid production caused by a gonadectomy. This treatment continues throughout the patient's life, as disruptions to the body's natural hormone production may result in bone loss. In youths, treatment plans may suggest the use of puberty blockers to temporarily prevent the development of secondary sex characteristics, which provides a reversible option should the patient decide they do not wish to undergo the next stage of transition (Coleman et al. 2022). For those who wish to continue their transition, the next step is the administration of Gender Affirming Hormone Treatment (GAHT) to initiate the puberty of the patient's desired sex (Coleman et al. 2022). During this time, bone growth returns to normal, bone maturation continues, and a growth spurt begins as the body undergoes puberty (Coleman et al. 2022). If individuals elect to undergo the recommended gonadectomy, they are prescribed long-term GAHT to compensate for the lack of natural sex steroid production.

While GAHT has a long history, puberty blockers are a more recent development. Puberty blockers are a pharmaceutical way to prevent the progression of puberty in adolescents by halting their production of estrogen or testosterone, providing a reversible initial step in transitioning. By administering GnRHa (gonadotropin-releasing hormone analog), doctors are able to suppress the progression of puberty (Coleman et al. 2022; D'hoore and T'Sjoen 2022). Puberty blockers have become a controversial aspect of transitioning in recent years with the *New York Times* publishing articles about medical concerns for individuals who take the medication (Twohey and Jewett 2022; Ghorayshi 2023). Concerns about the age of competency and a teen's ability to consent to the long-term effects of puberty blockers are prominent in these articles, but these are ultimately outside the scope of this thesis. Instead, this section briefly reviews key research on puberty blockers, BMD, and the effects of puberty blockers and GAHT on the skeleton.

Van Kesteren et al. (1998) conducted one of the first studies on bone mass effects of hormone treatment when they surveyed 20 male-to-female and 19 female-to-male transgender individuals who averaged around 25 years old. Individuals in this early study were treated with GAHT in preparation for a gonadectomy, after which they were treated with lower long-term levels of GAHT. BMD was measured in patients upon intake to establish a baseline throughout treatment, they were then measured again at 1 year after 28-63 months (Van Kesteren et al. 1998). This study was conducted because researchers at the time were aware that sex steroids affect bone mass, but did not understand the long-term effects pharmaceutical methods would have on their patients. The results from this study found that female-to-male patients had a gradual decrease in BMD which researchers attribute to an insufficient androgen given to those who had undergone an ovariectomy (Van Kesteren et al. 1998). Male-to-female participants had signs of lower bone turnover, but no bone loss occurred, which researchers attributed to the supplemental hormones. The researchers concluded that cross-sex hormone therapy does affect bone mass and bone metabolism, and called for more research in the area (Van Kesteren et al. 1998).

Another early longitudinal study to follow adult transgender individuals looked at bone mass density at 3 months and 12 months after beginning cross-sex hormone therapy (Haraldsen

et al. 2007). 33 transgender individuals in Norway between 1996 and 1999 were included in this study with an age range of 20-46 and compared to a control of 122 individuals (Haraldsen et al. 2007). X-ray absorptiometry (DXA) was used to measure BMD in participants, as well, biochemical markers of bone metabolism were measured. The study found that male-to-female patients had 50% lower values for the bone formation and resorption markers than the control did at 12 months. However, female-to-male patients had elevated bone formation markers over time (Haraldsen et al. 2007). This study is limited by the short period it followed its participants, but emphasized the need for longitudinal studies to see how these trends continue to progress with ongoing treatment.

Klink et al. studied the effects of hormonal treatment on bone mass using data collected at their institute, VU University Medical Center in the Netherlands, where children with gender dysphoria can receive GnRHa treatment from the age of 12. If at age 16 individuals still desire a sex change, they are prescribed cross-sex hormones to achieve the desired gender-affirming results (Klink et al. 2015). The study used retrospective data of individuals who had undergone the full course of treatment (including a gonadectomy) and whose bone mass (areal bone mineral density – aBMD – of the lumbar spine and femoral neck) had been measured at three times: at the start of GnRHa, at the start of CSH therapy, and at 22 years of age. The researchers divided their study sample into transwomen and transmen, noting other recorded measurements at these three times including their height, weight, BMI, pubertal staging and hormone levels. They note that one limitation of the data is that due to the nature of this retrospective study diet and physical activity could not be considered since they were not monitored; these are two major factors in bone mass density. The transmen and transwomen treated had lower BMD z-scores after GnRHa treatment than before; although both groups had an increase in bone mass following

GAHT treatment, it was still not equal to the typical score in individuals who had not undergone GnRHa therapy (Klink et al. 2015). They conclude that their results suggest a potential increased risk factor for fractures later in life, a topic for future research.

Jospeh, Ting and Butler (2019) conducted a longitudinal study of DXA scans of youths aged 12-14 undergoing GnRHA treatment. Their data set consisted of DXA scans done one year apart for 70 transgender youths, with a third scan at the 2-year point for 31 of the participants. This study was the largest to date to measure the skeletal effects of GnRHA on a longitudinal basis. Comparing their results to the size adjusted standards of BMAD developed by Crabtree et al. 2017, the following pattern was outlined. BMD z-scores showed a progressive decrease following the implementation of GnHRa treatment, most rapidly occurring over the first year. However, participants did not have any significant changes to absolute BMD scores. The authors suggest this indicates that sex hormone withdrawal during GnHRa treatment does not cause a direct loss of BMD but does cause a delay, leading to low bone mass compared to untreated individuals of their age group (Jospeh, Ting, and Butler 2019).

Vlot et al. (2017) argued the opposite when they looked at 34 female-to-male individuals and 22 male-to-female individuals who were treated with GnRHa triptorelin followed by crosssex hormone therapy at incremental doses when the individuals reached 16. The study measured bone turnover markers and bone density in the lumbar spine and femoral neck. Researchers found that during GnRHa treatment, there was a decrease in bone turnover accompanied by decreased BMAD z scores, though absolute BMAD did not decrease. 24 months after the initiation of cross-sex hormone therapy, BMAD measurements indicated that while bone mineral density increased from the initial scores taken following the halting of puberty, individuals still had low z-scores for their age group (Vlot et al. 2017). Vlot et al. concluded from their data that using GnRHa to suppress puberty leads to a decrease in bone turnover and lowered bone mineral density for age. This situation is improved by the use of GAHT, but GAHT does not return individuals to the average bone mass density for their age within a two-year period. These researchers advocate for the continued yearly DXA scans to monitor BMD.

As individuals continue to transition and have regular DXA scans during GnHRa treatment and subsequent GAHT therapy, more retrospective studies can occur. Until then, we can conclude that GnHRa does influence bone mass density. However, this is not usually a matter of absolute bone density loss, but of decreasing z scores reflecting delayed bone growth, meaning they have lower BMD than others of their age group who have not undergone GnHRa therapy. Additionally, individuals are treated with supplementary hormones (GAHT therapy) during and after transitioning. This triggers puberty and then maintains the desired transition results, but it also prevents further loss in BMD for age and partially or completely reverses losses during GnHRa therapy. The effects of puberty blockers on the skeleton are important to understand for insight into current gender affirming care. However, they do not offer a useful tool for forensic identification. Given the many factors involved in bone mass, any bone mass effects of these treatments cannot serve as a reliable clue to transgender identity.

4.1.2 Oral Contraception

Puberty blockers and cross-sex hormone therapy are not the only BMD-affecting hormone treatments used by transgender individuals. Some male-to-female individuals will use oral contraception, also known as birth control, for its feminizing side effects. Oral contraception is so common that the CDC reports approximately 9.2 million women using it in the United States in 2015-2017 (Daniels and Abma 2022). As a result, the effects of oral contraception have been studied in depth by clinicians, including research on BMD loss and increased risk of osteoporosis in its users (Almstedt, Shoepe and Snow 2005). Oral contraceptives are easily accessible to adolescents in Canada through walk-in clinics that do not require individuals over 16 to have any kind of adult accompany their appointment, with pharmacists in Alberta, Saskatchewan, Quebec and Nova Scotia being able to prescribe forms of contraception without consulting a primary physician (Di Meglio and Yorke 2019; Soon et al. 2021). As a result, oral contraception can also be taken by individuals who wish to regulate their hormone levels themselves to appear more feminine and induce side effects like the growth of breast tissue (Idrus and Hymans 2014; Barakat et al. 2021). This practice has been portrayed in popular television shows like *Grey's Anatomy* (S. 2 E. 13, Yu 2006) and *Glee* (S. 4 E. 19, Murphy and Falchuk 2013), and has also been discussed in blog posts on why individuals choose to take this route despite the health risks (Amato 2016; Hoyt and Kirkpatrick 2022). Literature on oral contraception and bone mass is a large area which has been well reviewed, the following section will briefly consider a few of these reviews.

Lopez et al. 2014 conducted a review of 19 randomized controlled studies of oral contraception and bone mass in various languages with a wide variety of results. Most of these studies looked at two hormonal contraceptives and compared them. Lopez et al. (2014) summarized that while some of the medications had no effect on BMD, other formulas had more effects. Overall, they concluded that the effects of lifestyle and diet are too variable between individuals to assess associated fracture risk from oral contraception alone. Randolph and Karvonen-Gutierrez (2021) provide a more recent review and found the majority of studies found either no effect or a small positive minimal effect of oral contraception on BMD.

Ricker et al. (2024) is a United States military review studying the effects of various forms of contraception on bone health. They found that depending on the study in question, the

use of oral contraception resulted in increases, decreases, and no change in BMD in study participants. This aligns with previous findings from the other reviews cited here that found oral contraception can have varied effects on BMD and bone turnover. The application of this data to the United States Military is for the use of oral contraception by female soldiers. The data from this review is intended for women in the military with regulated diets and active lifestyles leading the authors of the report to conclude that effects on BMD will likely be minimal in this group but this may not be the case for all users.

Based on this brief review, the bone mass impacts of oral contraception when used as a method of gender affirmation are too poorly known to be a potential factor for forensic anthropologists to use in identification. The studies of oral contraception and bone mass have been done on cis women; biological males have a different hormonal baseline as well as differences in bone size and bone mass, meaning that more research is needed on the extent taking oral contraception can have effects on the male body, including effects on BMD. As well, the variable effects found in cis women suggest that the effects of oral contraception would also vary between male-born individuals.

The results of this review suggest that conducting further research on bone mineral density focusing on its utility for identification as a skeletal indicator of a transgender individual would not be an effective use of resources. This is because transition-induced changes to bone mineral density appear to happen on a relatively small scale and are often countered by cross-sex hormonal therapy. Any changes to the bone would not likely be noticeable given the number of factors that can impact bone mineral density in life. This makes reduced bone mineral density a non-starter for the identification of trans individuals. The research done by clinicians is currently interpreted as confirming that while some gender-affirming care can cause lower BMD there is

not a large risk to individuals, meaning that this change is not going to present substantially in the skeletal record.

4.2 Surgical Interventions

Surgical intervention is another method of altering the physical body to seek gender affirmation. These procedures take a variety of forms but often focus on the secondary sex characteristics and cultural ideas of feminine and masculine presentation to help an individual achieve their desired appearance. Surgical procedures vary in their intention and extent; some are purely cosmetic (often the procedures performed on the face or the use of breast implants) while others have larger medical implications like the removal of the breasts or the addition of a penile implant. It is important to note that surgery is not a requirement for transgender identity, rather it is a method through which those with economic means can choose to affirm their identity.

Facial feminization surgery is the common name given to surgical procedures where individuals opt to have more masculine cranial skeletal traits surgically minimized. The scope of this surgery can vary from patient to patient, depending on their existing facial structure. These procedures encompass both bone and soft-tissue operations as the goal of these surgeries is to cure gender dysphoria in life, rather than to specifically alter bone structure.

Current approaches to facial feminization surgery date back to the 1980s with the work of Dr. Douglas Ousterhout Dr. Ousterhout published a series of works beginning with the 1987 "Feminization of the Forehead" where he considered anthropological work on sex determination of this part of the face (see Chapter 3.2) and traditional cultural conceptions of feminine and masculine to develop treatment protocols unique to each patient in which supraorbital contours would be reduced to achieve the desired feminine appearance. He divided his subjects into three groups based on the treatment required and the initial forehead contour. Group 1 was individuals

with normal (meaning minimal) bossing of the supraorbital area in a way that could be described as flat or slightly anterior-projecting supraorbital rims, a small torus and thick bone of the frontal sinus (704). Group 2 was individuals with relatively normal or slightly anteriorly placed supraorbital rims but combined with thin frontal bossing over the sinus (704), and Group 3 was individuals with excessive anterior projection of supraorbital bossing, so much so that reducing the bone would not be possible without entering the frontal sinus (705). Surgeries for both Groups 1 and 2 were considered minor procedures under local anesthesia (Ousterhout 1987). Dr. Ousterhout used this study to create standards of forehead protrusion and practices of filling down the supraorbital rim to appear more feminine.

These procedures were all clearly outlined by Ousterhout, including where surgeons should cut into the skull, in his 2010 book called *Facial Feminization Surgery: A Guide for the Transgendered Woman*. The book outlines what procedures are available, the risks involved, and images of how procedures are performed in consumer-friendly ways to inform the transgender reader planning to undergo these operations (Ousterhout 2010).

In addition to surgical interventions on the forehead, reduction of the mandibular angle is the other mainstream procedure that would be evident in skeletal remains. This procedure is most efficiently performed on individuals with moderate or mild mandibular angles; for those with larger mandibular planes, an out-cortex splitting is necessary (Hann and Kwon 2022). What this means is that in some patients, the



Figure 1 A) is an I-shaped osteomy. B) shows a L-shaped osteomy. (Hann and Kwon 2022)

procedure on the mandible is accompanied by surgery on the zygomatic. In these procedures, shown in Figure 1, a piece of the zygomatic is removed to change the overall angle of the bone to create a more feminine appearance while maintaining a proportional appearance to the cheek and jaw angle (Hann and Kwon 2022). To reduce the chin, a similar osteotomy is performed in which the angle of the chin's protrusion is measured from the antegonial notch to the midpoint on each side and a small vertical piece of bone is taken from the chin to increase the angle while reducing the overall surface area of the chin (Keyhan et al. 2015).

After studying rhinoplasties and facial reconstruction under Dr. Paul Tessier in the early 1970s, Dr. Ousterhout applied his new knowledge of reconstruction to facial masculinization surgery. These surgeries are easier to identify as they often are accompanied by bone grafts,

screws, and prostheses, all of which can be evidenced in the skeletal record. To masculinize the forehead, surgeons look for ways to increase the bossing around the frontal sinuses and increase the superior lateral orbital angle. This is almost the reverse of the procedures done for facial feminization surgery in terms of desired appearance, but the actual operation is very different.



Figure 2 Forehead Masculinization in Ousterhout 2011, P. S11.

Methyl methacrylate is applied to the forehead and then held in place by the locking holes seen in Figure 2 (Ousterhout 2011). The methyl methacrylate is then cooled and contoured; this stage can be repeated if a more intense look is desired (Ousterhout 2011). For chin masculinization surgery, a forensic anthropologist would see evidence of screws and plates as the chin is cut, repositioned in 2 segments, and then plates and screws are used to stabilize the chin in place Figure 3 (Ousterhout 2011). This technique has been used to augment the chin in 3 directions and is often referred to as a sliding genioplasty. The space created by the repositioning can be filled with a mixture of hydroxyapatite, the patient's blood, and Avitene to create a stable segment with the potential to have some subperiosteal calcification (Ousterhout 2011). From the images of the procedures provided by the surgeons, it appears that forensic anthropologists should be able to observe cut marks, metal equipment, or the presence of methyl methacrylate on the skull which can be included in their

reports.

In 2017 Dr. Ousterhout and a team of plastic surgeons at Deschamps-Baly Clinic conducted research in order to find a way to create the appearance of an Adam's apple in female-to-male transgender individuals. The surgery they developed involved harvesting the patient's rib cartilage and repurposing it into an Adam's apple (Deschamps-Baly et al. 2017). Should forensic anthropologists encounter rib cartilage grafted onto the decedent's thyroid cartilage it is likely that the odd location of the cartilage fragment would be noted, but if the anthropologists in question did not know these procedures, they might not realize that the graft could be a sign of facial masculinization surgery.



Figure 3 A Sliding Genioplasty presented in Ousterhout 2011, p. S12.

A cartilage graft is not the only surgical implant a forensic anthropologist may encounter in a transgender decedent. Three other major implants are common for gender affirmation: Medpor implants, saline/silicone implants, and penile implants. In previous medicolegal cases, implants have made an appearance in court proceedings and helped in the conclusive identification of remains as the serial number associated with the implant was used for individuation. The potential importance of implants to identification is not news to forensic scientists. Applications in the published case literature include using manufacturing stamps to backtrack to the hospital the individual received an implant in (Khartade and Meshram 2020), comparing dental records to crowns and other dental prostheses recovered with remains (De Angelis and Cattaneo 2015), and identifying individuals from the prosthetic limbs present in partially incinerated remains (Simpson et al. 2007). Implants may provide anthropologists with a serial number that is tied to the decedent's lived identity and can be used as a method of identification. It is easy to ignore the details of these surgeries in favour of this serial number, but each procedure is unique and affects the body in different ways.

The Medpor implant is a medical-grade polyethylene implant that has a more solid structure than silicone. Because of this, it is opted for during some procedures where the goal is to mimic bone. Mandibular angle masculinization can be done using a silicone implant, a Medpor implant, or a bone graft, but Medpor is the most common and recommended method due to its quick healing time compared to a bone graft and solid consistency making it appear more natural than silicone (Ousterhout 2011). The procedure is done by splitting the mandible and inserting the implant. This forces the pterygomandibular ligament to stretch over the new implant causing a more natural appearance. A bone graft, when placed under the lateral aspect of the mandibular angle, will often be resorbed and therefore is only used when the patient specifically requests that they do not want an artificial implant. The results of a bone graft are also less dramatic as the angle was limited to 6mm of augmentation at the time Ousterhout's book was written (Ousterhout 2011). Saline and silicone implants are surgically placed within a patient at three standard locations, the breast, hip, and buttock. The most common use of silicone and saline implants is in breast augmentation. Breast augmentation refers to the overall use of surgical methods to change the appearance of the breast (Ramachandran 2008). This can include implants or cosmetic surface changes to the nipple or temporary injections to make breasts appear fuller. Modifications can also include the removal of the breasts (a mastectomy) or reduction depending on the desired appearance of the patient (Ramachandran 2008). Surgical implants, made of saline or silicone, also do not affect skeletal anatomy, leaving behind no skeletal markers, but they do leave behind the physical implant itself. Silicone implants are very durable and resistant to decomposition and are imprinted with a serial number (Lanzarotta and Kelley 2016). This serial number is recorded and if found with remains, can help to determine who the individual was regardless of their gender. This is the same for hip and buttock implants; if the remains are intact, the serial number on the implants can be used for identification.

There are three standard penile implants currently being used, which vary depending on the individual's needs and desires: 3-piece inflatable, 2-piece inflatable, and non-inflatable semirigid implants. A 3-piece inflatable device has three parts: "a pair of corporal cylinders, scrotal pump, and abdominopelvic reservoir" (Krishnappa, Fernández-Pascual, and Martinez-Salamanca 2022, 8). This implant is desirable for individuals who wish to simulate erection when the scrotal pump sends the reservoir fluid to the head of the corporal cylinders (Krishnappa, FernándezPascual, and Martinez-Salamanca 2022). The 3-piece implant is also the only model that can be inserted both peno-scrotally or infra-pubically (Krishnappa, Fernández-Pascual, and MartinezSalamanca 2022). The 2-piece inflatable implant has a pair of corporal cylinders and a scrotal pump, but not the abdominopelvic reservoir, rather the fluid is freeflowing inside the pump. This device works by squeezing the pump to move saline to the cylinders (Krishnappa, Fernández-Pascual, and Martinez-Salamanca 2022). The last kind of implant, a non-inflatable implant, is limited to only one or two positions as it is a simple rod that can be concealed within the body. It is important to note that the presence of one of these implants does not mean that the individual is necessarily transgender, this procedure is also, and often, performed in biological males suffering from a penile insufficiency (Krishnappa, Fernández-Pascual, and MartinezSalamanca 2022). That being said, the presence of a penile implant will also be accompanied by a manufacturer number that should have been documented when the procedure was performed. If recovered, this number can be used to help identify the individual. If the number is obliterated, the presence of an implant should be noted and presented in the case report; this information can then be used in tandem with other potential transgender skeletal markers.

Surgical body modification can take a variety of forms. Descriptions of the ones outlined in this section all suggest they would be visible in the skeletal record, but more research is needed to confirm this. Implants and surgical equipment remaining in the body can provide clear indicators of human intervention. Surgical scarring on the other hand does not have the same assurances, instead, it may be confused with pathology due to the remodelling nature of bone. One case has been presented where a Computed Tomography (CT) scan was taken of a living transgender woman and discussed in terms of its forensic implications. This offers a chance to explore how obvious some procedures may appear and what they could be confused for.

4.2.1 Case Study – Flaherty, Byrnes and Maddalena (2023)

Flaherty, Byrnes, and Maddalena (2023) were approached by a 38-year-old transgender woman, called 'Patty' for privacy, who has had 12 procedures, listed in Table 1, as part of her facial feminization surgery journey. Patty approached the University of Nevada's Forensic Anthropology and Bioarchaeology Laboratory to learn more about the identification of

transgender people and how the computer system FORDISC evaluates sex. Patty returned to the lab with a 3D replica of her skull created using computed tomography (CT) taken in 2019 (Flaherty, Byrnes, and Maddalena

2023). The craniometrics of Patty's skull were interpreted by FORDISC as indicating a male, despite the substantial work that has been done to feminize her appearance. The research team compared these findings to the only other published



Figure 4 Patty's 3D-printed skull. (Flaherty, Byrnes, Maddalena 2023)

study on transgender women and FORDISC which analyzed the pre- and post-operative 3D models of eleven transwomen who had facial feminization surgery; that study also found that the individuals were all classified by the program as male (Rogers, and Deschamps-Braly 2020; Flaherty, Byrnes, and Maddalena 2023). These studies show the inability of the computer program FORDISC to distinguish between transwomen and cis men even when facial feminization surgery has taken place, but the images the studies provide also allow us the

opportunity to look at non-metric traits that are associated with the procedures and determine which of the operations are likely to leave traces visible to the observing anthropologist's eye. Although I have no experience in plastic surgery, I can speculate using the data presented by Flaherty et al. 2023 in Table 1 and Figure 4 as to what procedures caused the abnormal markings on the skull and how they might be misinterpreted.

Patty had two procedures focused on the forehead region, orbital rim shaving and forehead contouring. From the procedure descriptions, provided in Table 2, I speculate that the visible line above the supraorbital margin that is seen on both left and right is the result of a cut line caused by the forehead contouring which is described as removing a section of the glabella, reducing it or replacing it entirely with new bone. This line provides an example of situations where it would be useful for anthropologists to have more knowledge about healing and remodeling following plastic surgery. The line is more remodeled and less visible on the right side of the temporal than on the left. The symmetrical appearance of these lesions would strongly suggest some sort of cultural modification to the examining anthropologist, but the nature and extent of the modification might only be clear after further research. The forehead lesions might also be misinterpreted as indicative of an extreme sinus infection (Buikstra 2019).

The mandible has a distinct contour line in image D; this could be a result of the sliding genioplasty or the gonial angle reduction that Patty had performed. The two procedures occurred a year apart making it impossible to tell more information about which procedure caused which markings without further research. Images A and C show what appears to be a bone spur on the right side of the mandible. When compared to the left side of the mandible, there is a noticeable lack of symmetry that could be included in the individualizing factors of an osteological report. With more research on specific chin procedures and their healing, we could gain more insight

into why the marks of the procedure are not symmetrical, how these two procedures varied, and the role of bone remodeling following the procedure to determine if that is indeed a bone spur or an angle left by reducing the mandible.

The cheeks provide the most obvious sign of cultural modification by presenting something unique: a symmetrical series of small bony circles along the zygomatics. Patty had a cheekbone reduction followed by a reconstruction 4 years later. Although I could not find details on these procedures, I can only speculate that the circles visible along the cheekbones are the result of a reconstruction in which additional bony material was added. The symmetrical appearance of these markings and their uniqueness compared to what occurs in natural human variation make this procedure the most easily identifiable as a cultural modification. The visible additions and illustrate a need for more research on the materials surgeons use in their procedures to build up bone contours.

Patty's nasal bones also feature an unusual contour and irregular surfaces resulting from her three rhinoplasties. Without knowledge of Patty's history of rhinoplasty, the contour and additional bone growth could be confused with a history of breakage and remodelling. The lack of knowledge about healed rhinoplasties creates its own issue in identification, without previous knowledge of the procedure and its healing process can create conclusions not just in the identification of transgender women like Patty, but the many cisgender individuals who have received the procedure.

The procedures Patty underwent make her skull distinctive, and while Flaherty et al. (2023) found that they did not alter the FORDISC sex identification, there are multiple signs of surgery that an observing anthropologist could note. Patty's case is unique, as few people can access the number of cosmetic procedures Patty has, but when looking at all of the skull

modifications together, it becomes clear there is a level of cultural modification at play, even without knowledge of plastic surgery. Moving forward, learning to identify these features in isolation, and becoming familiar with the healing patterns and the procedures and products used by surgeons will help prevent confusion when these traits are observed.

Table 2 History of Facial Feminization Surgery (FFS) procedures undergone by Patty (Flaherty, Byrnes, and Maddalena, 2023).

Procedure	Year	Description
Primary Rhinoplasty	2002	Alters the size and shape of the nose, including the nasal bones.
Secondary Rhinoplasty	2003	Alters the size and shape of the nose, including the nasal bones; alters areas of concern from previous rhinoplasty procedure.
Tertiary Rhinoplasty	2009	Alters the size and shape of the nose, including the nasal bones; alters areas of concern from previous rhinoplasty procedures.
Sliding Genioplasty	2010	Reshapes and realigns the chin region.
Tracheal Shave	2010	Reduction of the thyroid cartilage (Adam's Apple).
Gonial Angle Reduction	2010	Reshapes the width and flare of the gonial angle of the mandible, generally reducing the broadness of the overall jaw.
Upper Lip Reduction (Lip Lift)	2010	Reduces and lifts the upper lip.
Orbital Rim Shaving	2010	Shaving, filing, and/or general reconstruction of the orbit.
Browlift	2010	Raises eyebrows to a desired position on the face.
Forehead Contouring Type 3	2010	Reduces the overall size of the glabella region on the frontal bone by removing a section of bone, reducing the bossing, and replacing the bone.
Hairline Lowering	2010	Moves the hairline and associated soft tissue forward on the face.
Cheekbone Reduction	2014	Shaving and contouring of the zygomatic bones to reduce the width of the face
Cheekbone Reconstruction	2018	Shaving and contouring of the zygomatic bones to reduce the width of the face; alters areas of concerns from previous procedure(s).

Like the cultural modifications seen in archaeological remains, surgical interventions reveal details about individual lives. With future research, specificity in identification can be increased through better knowledge of these procedures. Some plastic surgeries include a prosthesis with a serial number, but when that is not the case, there is very little for forensic anthropologists to use in their assessment. Symmetrical and unusual lesions, like those on Patty's cheekbones, provide an obvious indicator of a cultural modification but would be difficult for an anthropologist to interpret without knowledge of the surgical causes of what they are seeing. Other procedures may be easily confused for trauma when not presented in a symmetrical fashion. In cases like Patty's where the right side was more altered than the left, knowledge of how the skeleton responds to plastic surgery would help to understand the lesions more. While not every transgender individual will undergo plastic surgery, this area of research will potentially benefit those who do as well as the larger cisgender population.

4.3 The Archaeology of Castration

One example of major hormone alteration that can be seen in the archaeological skeletal record is castration. Unlike modern transgender individuals, castration was historically often not a choice made by these individuals, rather it was associated with the role they served in society. The handful of archaeologically documented castrates provide insight into what happens to the skeleton if a dramatic alteration to hormones, in this case, a reduction in testosterone production, occurs. Testosterone is the primary androgen in men, 95% of which is secreted by the testes (Mohamad, Soelaiman, and Chin 2016). Testosterone has a direct impact on several aspects of bone metabolism and tends to promote bone formation (Mohamad, Soelaiman, and Chin 2016).

Castration is defined as any action, be that surgical, chemical, or any other means that results in the loss of testicular function accompanied by a dramatic reduction in testosterone production (Eng, Zhang, and Zhu 2010). The magnitude of the effect is influenced by the age at which castration occurs, as the younger an individual is at castration, the longer epiphyseal fusion is delayed resulting in individuals of above-average height (Eng, Zhang, and Zhu 2010). Evidence of castration on the skeleton has been studied throughout the world, beginning with German scholars Tandler and Grosz (Tandler and Grosz 1909; 1910; Zanatta et al. 2016).

They studied members of a living group called the Skoptzy who reported being as young as 10 years old when they were castrated. This castration was the result of a radical Christian idea that if the number of Skoptzy castrated reached 144,000 they would trigger the second coming of Christ (Wilson and Roehrborn 1999). This idea led the Skoptzy to castrate children and prisoners throughout a large area of Russia into Rumania and Bessarabia; however, only medical examinations of adult Skoptzy castrates have been recorded. Patterns of kyphosis in the spine were observed in multiple participants and x-ray revealed thinning of the bones of the skull in the adult Skoptzy (average age 64), who reported being castrated at an average age of 18.

Another country where castration is tied to religion but in a very different way is in India. Unlike the Skoptzy's practice of castration as a religious sacrifice while the castrated individual remained socially male, in India, there is a longstanding tradition of biological men who are castrated and live as women, considered a third gender known as Hijra (Nanda 1986; Lal 1999; Hinchy 2019). This varies from other cultures where eunuchs were still considered to be male; in India, these individuals occupy a different social space and undergo a transformation of gender, more similar to the modern-day transgender individual. The term Hijra itself has a complicated and colonial history and has become the overarching term used to refer to castrated individuals by Indian elites and colonial officers (Hinchy 2019). Despite this, the term is still used today in Indian communities, but the practice has become less common in the post-colonial era as Western ideas about gender were imposed on India along with other ideologies by the British Imperial Army (Porter and Louis 1999; Brown and Louis 1999; Al-Mamun et al. 2022). Hijra were once thought of as teachers and disciples but in the post-colonial era they have become shunned, and synonymous with prostitution (Hinchy 2019, 1; Nanda 1999). "Hijras were apparently effeminate and unable to perform strenuous labour, while such 'eunuchs' were

assumed to be incapable of moral improvement due to the physical changes castration produced." (Hinchy 2019, 11). Colonial officials would use the term eunuch interchangeably with Hijra only noting the difference in their way of dressing; Hijra dress in traditional women's clothing and represent a sacred third gender, rather than the term eunuch which just refers to a castrated male. Unfortunately, the process of becoming a Hijra was not always consensual with individuals as young as seven being kidnapped, castrated, and initiated into the world of the Hijra (Hinchy 2019).

One of the primary reasons for individuals to be castrated in the West was for social roles like serving high-born women, working in harems, or religious positions; this is believed to have begun in the late Roman Empire and continued through the Byzantine and Ottoman Empires (Wilson and Roehrborn 1999). Wagenseil examined 10 surviving Ottoman court eunuchs in 1927 and conducted an autopsy of another eunuch who died of presumptive typhus. After the autopsy and 3 additional examinations, they obtained the anthropological measurements and xrays of 4 eunuchs' skulls (Wilson and Roehrborn 1999). These individuals were castrated in service of their role within the courts as harem keepers and had lived for an average of 30 years following their castration, meaning most were castrated before puberty (Wilson and Roehrborn 1999). Throughout their review, Wilson and Roehrborn noted kyphosis of the spine in the living Skoptzy and failure of epiphyseal closure in the skeleton examined.

Farther east, the practice of castration began in China over 4,000 years ago and ended in the early 20th century (Eng, Zhang, and Zhu 2010). Wagenseil again had the opportunity to examine eunuchs, this time in a German hospital in Beijing, and found that of the 31 individuals he examined, the average time since castration was 38 years (Wilson and Roehrborn 1999). He observed that 20 of the 31 men had kyphosis of the spine which they suggest is connected to the development of osteoporosis (Wilson and Roehrborn 1999). Kyphosis is defined by an increasing forward curvature of the spine along the sagittal plane (Lam and Mukhdomi 2023). Kyphosis is often related to underlying osteoporosis and its accompanying fractures that result in disc fractures and the signature curvature kyphosis presents with (Lam and Mukhdomi 2023). In 1960 Wu and Gu examined 26 eunuchs living in Beijing, they found that the average age of castration for these elderly men ranged from 10-26 (Eng, Zhang, and Zhu 2010). Nearly half of the men developed gynecomastia resulting in the men being heavier than non-castrated men of the same population.

Due to the role castration played in the social roles of these individuals, their castration was often a choice made after the onset of puberty and the closure of the epiphyses. However, there are a handful of bioarchaeological studies of known castrated individuals. The studies provide insights into how the skeleton may change in individuals who halt hormone production without the supplementation of cross-sex hormones to maintain healthy bone mass. As detailed below, these individuals have delayed growth resulting in tall postures and visible epiphyseal lines, and low bone mass unique to castrated individuals.

The Castrati were opera singers from the end of the 16th century throughout the 18th, who were castrated at a young age to maintain their higher young voices (Zanatta et al. 2016). The Italian opera singer, Gaspare Pacchierotti (1740-1821), was one of the most famous castrati of his time, and when his remains were exhumed in July of 2013, his became the first castrato skeleton ever analyzed (Zanatta et al. 2016). Pacchierotti was described to be quite tall, and his stature was estimated to be approximately 191 cm using length of the femurs, tibiae, and humeri; this aligns with the remarks of the era that the "castrati" were particularly tall individuals (Zanatta et al. 2016). His skull featured visible fusion lines at the iliac crest epiphyses despite his

advanced age (Zanatta et al. 2016). After performing CT scans on Pacchierotti's spine, the researchers found indications of low bone mass (Zanatta et al. 2016). Pacchierotti's was the first whole castrato skeleton ever analyzed, but he was not the first to be excavated; Carlo Broschi, more famously known as Farinelli (1705-1782), was studied in 2006. His partial remains notably had an epiphyseal line present on the iliac crest, reflecting the fact that he was castrated at a young age (Belcastro et al. 2011; Zanatta et al. 2016). Similar to Pacchierotti, Farinelli was estimated to be 190 cm using the ulna and also featured notable signs of low bone mass (Belcastro et al. 2011).

Some researchers argue that Hyperostosis Frontalis Interna, or HFI, a disorder seen most often in middle-aged women, is another result of castration. HFI is the result of a hormonal disturbance that can present as the thickening of the frontal bone, excessive hair growth, epilepsy, and disturbance of the sex glands or gonads (Hershkovitz et al. 1999; Chaljub et al. 1999). A skeleton of a castrated individual curated in the Medical University Anatomical Museum, Lyon, France shows evidence of HFI (Reusch 2013). The Lyon Castrate, who died in his early twenties, appears to have been in the first two stages of HFI, often the stage associated with women in their thirties today (Reusch 2013). Farinelli also had indicators of HFI as his frontal bone had a thickened diploe and appositional bone adjacent to the sagittal sulcus (Belcastro et al. 2011). However, Pacchierotti did not show cranial table thickening and when Zanatta and colleagues examined radiographs from the Skoptzy, they found cranial thinning. Because HFI is a hormonal disorder typically seen in menopausal women, Reusch (2013) argues that when combined with other skeletal factors this could be used as an indicator of castration when looking at ancient remains; Zanatta et al. (2016) however argues that the variation of cranial thinning and thickening in the skulls of castrated individuals is too variable to associate with castration.

Reusch (2013) evaluated the Lyon castrate and the skeleton examined in 1909 by Tandler and Grosz and found that both had pelvises which appeared masculine in the morphology of their subpubic angle and sciatic notch but also had feminine-appearing features, including shape of the obturator foramina, which are triangular, a shape associated with females; males have a more large and oval shape (Ridgeway, Arias, and Barber 2008; Reusch 2013). The pelvis also featured posterior flaring of the ilia, resulting in a more typically feminine shape when examined from the superior view (Reusch 2013). This, combined with evidence of delayed epiphyseal fusion, begins to create expectations of the skeletal changes we should see across individuals who were castrated prepuberty.

Eng, Zhang, and Zhu examined two documented eunuchs excavated from a cemetery in Wutasi, Beijing in 2003. These individuals, one whose name was not preserved and the other whose tomb inscription identifies him as Huang Zhong, lived during the Ming Dynasty (13681644) a time when eunuchs were called one of two things, huanguan or taijian, indicating if they were castrated as juveniles or adults (Eng, Zhang, and Zhu 2010). This was done for one of two reasons, to raise men to be safer servants for royal women, or as a punishment for prisoners of war (Eng, Zhang, and Zhu 2010). The first burial, Burial M1, is of a male believed to be in his mid-30s at death (Eng, Zhang, and Zhu 2010). This individual had typically masculine sexually dimorphic features and full epiphyseal closure, indicating that he was castrated after undergoing puberty. The dentition shows three hypoplastic lesions on a left upper incisor, two on a right upper incisor, and one on a right upper molar, which Eng and colleagues argue was not the result of castration, but rather the result of the stressful social factors that potentially led to M1's castration, "to escape poverty, to serve as tributes, or as prisoners of war." (Eng, Zhang, and Zhu 2010, 114). Burial M1 had a number of degenerative changes in postcranial joints and the left mandibular condyle was more bulbous than the right which they suggest indicates temporomandibular joint disease (Eng, Zhang, and Zhu 2010). The age at death of Burial M2, Huang Zhong, was not as easy to determine. The epiphyses of the long bones were unfused or partially fused, making it seem as though this individual would be in his mid-to-late teens, but the presence of fully erupted third molars with evidence of slight wear, along with auricular surface age, indicate he was more likely between 20 and 24 at the time of death (Eng, Zhang, and Zhu 2010). Burial M2 also had some sexually ambiguous cranial morphology, though the pelvis was determined to be male in morphology and considered "relatively diagnostic considering his young age" (112). Burial M2, Huang Zhong, was estimated to be approximately 188 cm in stature compared to Burial M1's 179 cm approximate height using femur length. The authors suggest this is not unusual given that Burial M2's delayed long bone epiphyseal fusion indicates the individual was castrated before the onset of puberty (112). Burial M2 also had evidence of enamel hypoplasia but the authors again do not attribute this to the act of castration, rather the living conditions leading up to it as the crowns of the teeth would have formed before the individual was castrated.

Throughout the limited skeletal remains of castrated individuals available to researchers, a few patterns can be seen, the clearest and well-agreed-upon being a lack of epiphyseal fusion of long bones and the iliac crest as an indicator of castration before puberty. It is important to note that some of these castrated individuals analyzed lived in specific contexts that also could have affected their development, including the intensive breathing exercises of the castrati (Zanatta et al. 2016). Multiple individuals showed signs of kyphosis and had lower bone mass than others of the same age. Morphologically, these skeletons are still sexually dimorphic enough to determine they are male based on pelvic morphology, but there are some more typically female traits including outward flaring ilia and triangular obturator foramina (Reusch 2013).

While the archaeology and history of castration are different than the lives of modern transgender and gender-diverse individuals, the skeletal effects of delayed epiphyseal fusion for an individual's age combined with increased stature are consistent findings resulting from the prepuberty halt of testosterone production. However, as discussed in section 4.1, modern transgender individuals are typically treated with cross-sex hormones which will allow growth to continue on a more normal trajectory and prevent the poor bone mass found in eunuchs. While this area of research is not useful for transgender identification in practice, understanding how the abrupt halt of testosterone during development affects bone growth provides insight into why transgender individuals are treated with lifelong cross-sex hormones.

4.4 Conclusion

Like defining sex, the identification of transgender individuals is multifaceted and does not have any one answer. Modern pharmaceutical methods of transitioning do not appear to have any effect on bone development. The use of supplemental cross-sex hormones prevents the dramatic effects seen in castrated individuals whose puberty was halted, and most modern methods of transitioning are a gradual process designed not to disrupt the body's development, but rather to alter it to affirm an individual's desired gender expression. Surgical intervention holds more promise for identification but is limited by socio-economic factors preventing individuals from accessing treatments. Despite that, research in this area would still be beneficial for identification as plastic surgery becomes more accessible on a global scale to both transgender and cisgender individuals. Treating plastic surgery as a cultural modification, similar to how it is practiced in archaeology, would better allow forensic anthropologists to interpret what they are seeing and avoid misinterpretation. The research presented by Flaherty et al. (2023) shows that modifications are visible to observation and the potential to continue using CT to increase research in this area with living individuals.

Chapter 5: Interview Component: Methods and Results

After assessing the need for research in this area in Chapter 2, reviewing our understanding of sex and sex estimation in Chapter 3, and looking at potential skeletal identifiers of transgender individuals in Chapter 4, this chapter addresses how this knowledge relates to the field. I wanted to better understand how the proposed terminology changes, methods I researched, and current discussion around sex and gender fit into the medicolegal side of forensic anthropology, outside of the classroom and published literature. In developing my research questions, I looked to the existing literature and found a research project that surveyed forensic anthropologists on their opinions about the need for transgender-related research (Tallman, Kincer, and Plemons 2022). This survey asked participants about their experiences in performing sex estimation, casework experience with transgender remains, and student engagement in this topic. Responses to the survey reinforced the need for research in this area. This led me to wonder exactly what sorts of research could be done, and which directions would be the most beneficial to those who are assessing the decedent's remains. Making use of their survey results, I incorporated these and other questions into an interview script intended to get more in-depth answers than a survey could provide.

Interviews were conducted one-on-one over Zoom or in person with a small group of practicing forensic anthropologists. These anthropologists were of various levels of familiarity with the subject of transgender identification and recognized the desire for more research in this area. After our conversations, multiple themes emerged including a focus on terminology use, community engagement, identification methods, and where they see the benefits of future research given the status of research in this area thus far.

5.1 Study Design and Methods

To provide an informed perspective, participants all needed to be well-versed in the medicolegal system and to have practiced as forensic anthropologists within it. Knowledge of issues surrounding transgender identity was considered to be an asset in participant selection but was not an absolute requirement. Before contacting prospective participants, I secured my research ethics approval from the University of Alberta Research Ethics Board, under the project name "Beyond Binary Identification: New ideas of gender and how they fit into forensic anthropology", No. Pro00133307 (Appendix A). The initial participants I contacted had publicly mentioned their interest in the topic of transgender remains in forensic anthropology or had vocalized an interest in advocacy for disenfranchised groups. Further potential participants for my interviews were identified using snowball sampling, an approach that makes use of the connections of study participants to help with studying a hard-to-reach population (Johnson 2014; Goodman 1961). Participants were invited to share my letter of initial contact (Appendix B) with others they thought might be interested, and those individuals could contact me for more information.

In the end, I conducted four interviews, one in person and three using the online platform Zoom. The interviews took place between October 31, 2023, and January 19, 2024, with transcripts being returned to the participants for review within two weeks of the initial interview. Participants were at this time given the opportunity to remove, alter, or add to the transcripts but aside from clarifying the spelling of a last name, all participants opted to leave the transcripts verbatim. The interview data was stored on a secure Google Drive shared with only myself and my supervisor Dr. Sandra Garvie-Lok. Participants were given the option to remain anonymous or have their names associated with the following results: one participant, Participant 2, wishes to remain anonymous but is a practicing professional forensic anthropologist in Canada as well as an instructor at a university. The remaining three participants were Dr. Sean Tallman, an Assistant Professor at Boston University; PhD Candidate Ashley Smith at the University of Toronto; and Dr. Joel Cahn, an Assistant Professor at Trent University. This group is representative of a variety of locations across Canada with an additional American perspective from Participant Tallman in Boston and from Participant Smith's personal experience as an American practicing in Canada. These individuals, however, are a very small and non-random sample, and reflect the opinions and experiences of four individual forensic anthropology practitioners who are operating in the field at the time of writing; they are not intended to serve as a representative sample.

5.2 Results

The interviews began with questions on participants' positionality, asking them for their names, pronouns, and official titles. Participants were asked about their existing knowledge on the conversation of transgender decedent identification, with which all were in some way familiar; two participants been involved in the conversation themselves, Participant Tallman and Participant Smith (Flaherty et al. 2023; Adams et al. 2024). The interviews continued following the interview guide (appendix C) and as they progressed themes emerged, and additional questions were asked to explain and explore participants' answers in depth. The following sections present the key responses of participants organized by theme.

5.2.1 Using Cultural Context and Evidence

The Tallman, Kincer, and Plemons survey asked participants multiple questions about using cultural or contextual evidence like items of clothing at the scene of a crime in their interpretation of the evidence. This question had mixed responses, with a 50% split between those who see any information as useful and those who think only skeletal evidence is within the scope of the forensic anthropologist's analysis and report. My participants generally agreed that if they have access to that information then they would consider that an asset, however, due to the nature of some investigations, forensic anthropologists are not always brought onto a case until after the remains have been removed from the scene. One participant, Cahn, emphasized his concerns about over-emphasizing evidence and assigning meaning where there may not have been, giving the example, "they could be found with clothing, but they might have just been shopping for their sister or something" (Cahn Transcript 7:01-8:58). This was echoed by Participant 2 who voiced concerns about overstepping the boundaries of forensic anthropology in their investigation by looking at evidence outside of the skeleton.

This ties into the theme discussed in 5.2.3, participants' concern about overstepping their expertise and negatively affecting an investigation.

5.2.2 Plastic Surgery

In the case of gender-affirming surgery resulting in implants, while none of my participants had encountered this in their casework, they agreed that the presence of a serial number on the implant would suffice for identification. Participants agreed that clinicians rather than forensic anthropologists are the best to do further research in this area. Participant 2 argued that any additional knowledge about the decedent can be useful to an investigation and highlighted that increasing knowledge on plastic surgery, how it can present, and how it heals could prevent the examining anthropologist from confusing surgical scars with other pathologies. Participant 2 cautioned against using any one surgical procedure as an indicator of transgender status, rather advocating that evidence of a past surgical procedure is an individualizing characteristic regardless of its tie to gender expression and is therefore useful to the investigation and should be reported alongside individuating characteristics.

Regarding future research, I asked participants about plastic surgery as an identifying factor. Most participants responded that while facial surgical scarring may be evidence of genderaffirming surgery, only a small minority of the transgender community has access to those surgeries, so participants are hesitant to call surgical scarring a one-and-done identifying factor. In Canada and the United States, cosmetic procedures, which Smith points out gender-affirming surgery is considered to be, are inaccessible to most due to financial barriers and the long and draining process that getting coverage for these surgeries can require. Additionally, as Participant 2 noted, gender-confirming surgery is not a legal or social requirement for transitioning which also impacts the number of individuals who feel they need these surgeries to complete their transition.

Participants were asked about how they would currently report evidence of plastic surgery, and what research they would need to see in order to begin to feel confident in assertions about gender from this evidence. All participants agreed that they would note signs of cultural modification, including plastic surgery; what varied was where they would report on this information. They agreed that this information would be noted with other individualizing characteristics, but Participant Smith said she felt confident enough to include this information in the initial summary if she believed the surgery aligned with evidence of facial feminization surgery or facial masculinization surgery. She said she would include that information alongside the summary table in her report. While others felt the lack of research in this area made these interpretations challenging, Smith believes there is enough published now to allow a practitioner to explain the reasoning behind the report and explain why certain specific signs constitute an indicator of transgender status in life. All participants noted that it is not only transgender individuals who receive some of these surgical procedures, but when asked about this as a problem, they all made a similar point: the skeletal indicators are still evidence that the person had plastic surgery. While we may not be able to say if that surgery was done for gender-affirming reasons, we can still provide the police with the knowledge that the individual had a history of plastic surgery and providing that information to the police has the potential to reduce the number of missing persons. Future research may improve our ability to say what surgery was performed, further increasing this potential.

5.2.3 Concerns about Overreaching

Concerns about overstepping the boundaries of forensic anthropology were commonly voiced by my participants, not only in terms of the role the participants occupy in the investigation but also regarding identifying the decedent. Participants all voiced concerns about creating a transgender profile for individuals to fit into for identification, creating overgeneralization or a constricting image of what classifies someone as transgender. This concern was raised not only about items found at the scene but also the potential indicators created by plastic surgery. Participant 2 commented on the dangers of using some kinds of plastic surgery as an identifier of transgender status because of the prominence of these surgeries among cisgender wealthy individuals as well; examples of this are procedures like nose or breast augmentation. Participant 2 was also concerned that the current lack of knowledge about plastic surgery scarring on the skeleton could lead to additional confusion for both the anthropologist and the investigator interpreting the report, and warned those who encounter it to consider all the potential reasons surgical scarring may be present.

Participant Tallman and Participant Smith both noted that the majority of transgender individuals do not have access to gender-affirming plastic surgery. There is a substantial socioeconomic barrier preventing individuals from accessing non-emergent plastic surgery, even more so in the United States where Participant Tallman is located. When asked about images from Flaherty and colleagues' 2023 article (reviewed in Chapter 4) showing that there are indeed physical manifestations of facial plastic surgery on the skull, at least in living individuals, some participants did not see potential value in the future of this research. Participants generally agreed that future research on the identifying marks of plastic surgery could be pursued to inform forensic examiners but felt that it should not be done from the exclusive perspective of transgender identification. Rather, they felt that if examiners can establish that any individual has undergone plastic surgery, they can inform investigators that the decedent had undergone cosmetic procedures, increasing the likelihood of identification.

5.2.4 Considering Gender in Anthropology

Touching on other concerns, participants noted the complexities of gender in the Western world. Participant 2 voiced concerns about looking for transgender identifiers not matching the quickly moving ideas of gender in society. This was seconded by Participant Cahn who was concerned that creating guidelines would lead to implications of individuals not being 'transgender enough' if they cannot access gender-affirming care. It also does not take into account the number of individuals who identify as non-binary. Non-binary identity is often grouped with transgender identity as this too is a gender transition, however, it is a transition to a neutral state as opposed to a transition to the other side of the gender binary (Vincent 2020).
5.2.5 Terminology Use

These expressions of gender are more varied than our definition of sex, which leads into another interview question: how do my interviewees feel about proposed changes in the terminology used by forensic anthropologists to better reflect the complexity of gender? At the time of the interviews, one piece had been published on this discussion, arguing for a shift in terminology from "sex" to "assigned sex at birth, ASAB," on medicolegal forms to reflect that your sex is 'assigned' to you when the doctor writes it down on the official form at birth (Adams et al. 2023). This has since been followed by publications supporting the call for this change, including one article featuring two of my participants as co-authors (Flaherty et al. 2023; Adams et al. 2024). In my interviews, I asked all participants about their opinions regarding the shift from "sex" to "ASAB" to which I received mixed responses. Unsurprisingly, the two participants whose article was published shortly after the interviews were conducted both supported this call and mentioned their upcoming article. Participant Smith mentioned that sometimes at birth ambiguous genitalia are surgically altered to fit an assigned sex (Creighton 2001; Carpenter 2018; Dickens 2018). The altered genitalia do not always match the sexual organs of the individual or the gender identity they would come to display. My other two participants were concerned about the implications of using the term "assigned sex at birth" as they feared it could potentially add further implications and confusion to cases where confusion was not present before. Participant 2 was hesitant about using the term "assigned sex" because they felt it was not their place to assign sex to this individual. They said unless there are distinct morphological changes, it will not change the way they apply the Phenice traits to the remains. They continued by saying they were hesitant with terms like assigned sex because they said they were not qualified to interpret the individual's lived history this way. Dr. Cahn was concerned that saying "assigned sex at birth" would reinforce the idea that gender is tied to biology. These small

changes in terminology lead to connotations and inferences that forensic anthropologists should be aware of moving forward.

While discussing this subject, Participant Smith also mentioned her argument for the shift towards the term "osteological sex" as opposed to just "sex" when reporting sex estimations based on skeletal remains. She argues that "assigned sex" as a term used for living people is important because it acknowledges intersex individuals and those whose sex was ambiguous at birth and surgically assigned one sex or the other. Smith pointed out that on legal documents there may be downsides to using the term in case reports in situations where the term is not used universally across government documents, providing the example that in her home state of Tennessee, death certificates must reflect the assigned sex at birth, not the gender of the individual. She continued that when discussing the deceased, using the term "osteological sex" when reporting findings may help to alleviate some confusion and provide a very exact description of what forensic anthropologists are providing in their reports. "Osteological sex" also alleviates another issue highlighted by Smith who mentioned that biological sex actually may refer to four things - chromosomal sex, hormonal sex, anatomical sex, and osteological sex as discussed in Chapter 3.1. This shift in terminology would allow reports to reflect exactly what forensic anthropologists are able to interpret.

5.2.6 Increasing Assessment Confidence

In regards to reporting, I also asked my participants about what they would need to feel more confident in assertions of transgender status should they be presented with a case in which the decedent appears to have been transgender. Responses varied. Overall, all participants agreed that more publications on the matter would help them with identification, however even this varied. Participant 2 wished for more general research publications on topics like plastic surgery

from the perspective of overall improved identification for all individuals. Participant Smith said she feels confident in every report she sends because it is done to the best of her ability, so her confidence does not vary between reports, and it is not a matter of assurance in her reporting. Participant Cahn said, "I would need to have a skeleton that says one thing and the clothing or other assortments of things that says another to say that and even then, I'd couch it. I'd have to be careful." (Cahn Transcript 32:04-32:22). Participant Tallman said he would call in a plastic surgeon for a second opinion if he saw evidence of surgical scarring.

The majority of participants exercised caution around making identifications and desired more knowledge before making concise, affirmative, decisions because they worried about narrowing a suspect pool too far. The other participant, Smith, mentioned that her report is her professional opinion and when questioned on it she would draw upon whatever sources she could; this suggests that she also believes we would benefit from more publications as it would increase the number of studies she could cite in her arguments. This confirms what we already knew from Tallman 2022's survey, that there is a desire for more research projects in this area, but perhaps we would also benefit from more published forensic cases by those who do encounter these decedents.

5.2.7 Collaboration with Transgender and Gender Diverse Communities

Another question posed to participants came from the statistic that 97.3% of forensic anthropologists surveyed said that research on transgender individuals within forensic anthropology should be done in consultation or collaboration with transgender communities (Tallman, Kincer, and Plemons 2022). Going past studying the community, there is a desire to involve the community in all steps of the research process. When I asked my interview participants about the role they think transgender communities should play in forensic research, they agreed that working along with communities would be beneficial and they gave a variety of reasons. Participant Tallman noted that he thinks this impetus comes from the younger generation of anthropologists and their desire to incorporate social justice into their work; he continued that he believes older generations are more apprehensive about engaging in this research. Tallman said he believes this is rooted in anthropology's history. He mentioned that the work of anthropologists has historically led to negative impacts on many communities we have engaged with and points out that in many ways anthropologists are still complicit with the subjugation of some communities by not acknowledging the impact we have.

When participants were asked why they viewed community engagement as important, answers began to vary. Generally, they followed the same two lines of thought, looking at anthropology's history as a discipline and using cultural anthropology's method of working with communities and highlighting their voices rather than speaking for them. Participant 2 noted the famous phrase "nothing about us without us," a mantra of various rights movements dating back to the South African disability rights movement in the 1990s (Charlton 2000; Pfeifer 2022). When prompted about why they feel this way, Participant 2 referred to their experience working with Indigenous communities and the focus being brought to ensuring that we as anthropologists, as well as governing bodies, are listening to the voices of the individuals we are discussing.

One potential method participants were asked about was approaching community engagement by contacting local transgender alliances on a city level, but also working with groups like the Trans Doe Task Force in the United States. The latter is a group who identifies themselves as a task force that "finds and researches cases of LGBTQ+ missing and murdered persons, especially focusing on unidentified individuals who may have been transgender" (Trans Doe 2024). The task force shares the hashtag #**IfIGoMissing**, a social media outlet for individuals to share their – If I go Missing – folders including their personal details, which are then either in a secure location or entrusted to a loved one (Trans Doe 2024). The Trans Doe Task Force also adds a statement that addresses the concerns of multiple participants interviewed: what if it is not a safe environment to share that information, online or in the home? The Task Force has made their own database in LAMMP where people can include their information, personal details, how they would like to be represented, and identifying factors to be used if someone goes missing (Trans Doe 2024). This provides individuals who are not in a position to identify themselves openly online with the security of knowing their identity will be defended by the task force in the event that something should happen and a missing persons case is opened (Trans Doe 2024). The Task Force notes that while the police can also use the database LAMMP, they do not have automatic access to the part of the database where this information is stored; they stress that this information will only be shared after the Task Force has also made efforts to contact the individuals provided in the form. In exchange for sharing this information, the Trans Doe Task Force makes the promise "we will look for you" (Trans Doe 2024).

When participants were asked how we can begin to either better merge or ally the information the Trans Doe Task Force has with that held by the legal institutions forensic anthropologists work in, most simply the answer was allowing an indication on the form that this individual might fall outside the gender binary and therefore the police can reach out and share resources. As for how to begin to make that change, participants agreed with the suggestion of workshops or guest speakers at police training events but also indicated that it should not be entirely the Trans Doe Task Force's responsibility to facilitate this conversation. Participant Tallman noted that forensic anthropologists need to do a better job of communicating our role and the complexities of it with both the stakeholders and the general public. One way to change this that he suggested is training, but he also recommended having conversations with those we

work with in order to begin to reverse the oversimplifications in some of our practices. He went on to describe anthropologists being asked to summarize their work in a table or paragraph with one-word answers to make it easily understandable, the focus being on something easily entered into a database, a set of keywords. This again reinforced the argument that there is an issue with the current database system that working with communities and having conversations with stakeholders can potentially solve.

When discussing community involvement, Participant Cahn referenced historical fashion trends used by the LGBTQIA2SL+ community to communicate with each other covertly. He noted the value of involving the transgender community in investigations to help identify coded items at a scene. This kind of information sharing would be potentially challenging as Participant Cahn notes that "maybe the tensions are icy, that if you've been bitten or hurt before, you're going to be shy to approach it. So maybe there's some reservations there." Additionally, fashion communication was historically used to communicate queerness with others while avoiding the detection of the police, for example, the alleged use of a green carnation in the jacket buttonhole to indicate homosexual status in late 1800s Paris (Davies 2022), thus sharing similar coded information with the police may be a challenging hurdle to overcome. Aside from Davies' academic publication, discussion of these trends exists primarily in community spaces like blogs and in-person conversations, using flagging methods like the handkerchief code, a system using coloured handkerchiefs coded to sexual desires and fetishes (Peoples 2019). These methods of communication were designed to keep verbal communication minimal and secret from the police at a time when being gay was criminalized (QueerEvents Editorial Staff 2018).

71

5.2.8 Future Research and Conclusions

Participants agreed on the importance of including transgender communities in the research process and ensuring that our role in the conversation is advocacy for their voices, not speaking for the community. Similar to the suggestions of Adams et al. (2024), using the correct pronouns if known when referring to the decedent and correcting others involved in the investigation is one way anthropologists can advocate for decedents on an individual level. Participants also agreed that care around your language in reports and in discussions is crucial to respect the lived experience of each individual.

Participants all agreed that a specific focus on facial feminization surgery and facial masculinization surgery is not the best course of action in future research on signs of plastic surgery; rather, forensic anthropologists should focus more broadly on understanding how plastic surgery affects the skeleton. Until forensic anthropologists can better understand modern cultural modifications caused by plastic surgery, we should focus our efforts on this understanding and apply the knowledge to gender later. Participant Cahn did not see much value in this research because of the rarity of these surgeries and the likelihood that implants with serial numbers will accompany common plastic surgery procedures. He noted that any case involving a serial number, whether on an implant or a prosthetic, will be accompanied by a record tying the serial number to an individual, negating the need for additional work in the area. Participant 2 noted in their interview that plastic surgery is not just for transgender individuals, and is becoming more accessible and commonplace, potentially resulting in an overall increase in plastic surgery procedures. For this reason, there may be value in establishing potential patterns of surgical indicators. Participant Cahn commented that variation in procedures between surgeons may make research in this area challenging; this suggests the research will require a larger sample size with a variety of surgeons having performed the same procedures to compare the levels of variation.

To learn more about participants' opinions on the future of research, I asked about the potential for longitudinal studies of skeletal changes in individuals who transition. When discussing this topic, multiple participants mentioned the practical challenges of any long-term longitudinal study following individuals who are taking hormone therapy. As two participants pointed out, a longitudinal study on the required time scale is practically impossible at the student level given the time constraints of traditional doctoral or postdoctoral programs.

On the topic of bone mass research, participants all agreed with the conclusions drawn in Chapter 4 that the existing clinical studies in this area did not consider enough of the additional lifestyle factors that contribute to fluctuations in BMD. Three of the participants agreed that research in this area by forensic anthropologists is not feasible. They all argued that BMD is far too variable to evaluate consistently between individuals with differing habits and personal traits and that it would be impossible to determine what caused BMD loss in each individual if it did present. They also agreed that BMD loss would have to be very substantial to be noticeable at a level observable to forensic examiners, making the relevance of the studies to forensic practice questionable. One participant, Participant Smith, did see a future in similar work and mentioned her desire to do a post-doctorate on the microscopic composition of bone to see the effects of puberty blockers and other gender-affirming hormone treatments on individuals. Smith discussed her desire to pursue this research and look at the changes occurring to bone mass on a microscopic level from gender-affirming hormone treatment.

In the end, all of the participants agreed that BMD research was not a viable path for forensic anthropologists' research based on the preliminary findings of minimal BMD change, and that the limited resources of the forensic community would better be spent elsewhere. Participant Cahn strongly felt that work in this area would do more harm than good, spending valuable resources on something he believed could endanger transgender lives by spreading the ideology that gender is fundamentally tied to biology. He continued by saying that research may be valuable in this area for the living as part of monitoring transgender individuals' health, but from a forensic standpoint, this area of research should not be pursued. He went on to discuss his concern that it would limit the transgender profile to individuals who take puberty blockers. He said "You're going to restrict it to just people who can take the puberty blockers. It's not going to be effective! [...] You're not helping. You are harming because, at the very best, you're taking away resources and you are fooling people into thinking, and you're potentially engendering these ideas that transgender is fundamentally a biological thing that we can tell you are advancing it!"

The themes present throughout the four interviews provided additional insight into the research of the previous chapters. There is a desire to work with transgender communities and advance research that will benefit more than just transgender individuals, and working with communities aligns with the participants' voiced concerns about overstepping boundaries as forensic anthropologists in investigations. Participants also expressed concerns about assigning perceived sex to an individual, instead, they favoured the traditional osteological sex estimation methods which work on a sex continuum and note unique characteristics as identifying features. Moving forward guided by the opinions of those who are in the field performing identifications will provide the opportunity to maximize the effectiveness of future research.

Chapter 6: Conclusion

6.1 Summary

There are paths forward for conducting research to identify transgender individuals, but the future of this research needs to be deeply intertwined with the transgender community. If forensic anthropologists work with communities, it will improve our understanding of how the average transgender person transitions, whether they opt for surgery or just pharmaceutical intervention, the reasons behind these choices and potential socio-economic restrictions to transitioning, to help improve our understanding of what people are experiencing. The use of outside resources like the Trans Doe Task Force also shows potential future pathways to improve the identification of missing transgender persons. Working together requires levels of respect which importantly needs to translate to the descendant's care with proper pronoun usage and using your position to defend the descendant's right to proper identification as advocated by Adams et al. (2024).

Each core chapter in this thesis looked at a different area to help better understand of the process of transitioning and the role of the forensic anthropologist in a medicolegal investigation. Chapter 2 looked at why this research is important to conduct and a very brief look at some of the people who this work would benefit and why they are calling for it. Chapter 3 provides a review of the term sex and its use in developing forensic anthropology methods for sex estimation. In Chapter 4, the specifics of transitioning were evaluated when looking at the effects of pharmaceutical transition on bone mass as well as the looking at the archaeology of castration where pharmaceutical hormone care was not provided, to better understand the care modern transgender individuals receive. Chapter 4 also looked at the recent history of medical procedures designed for gender affirmation including implants and plastic surgery to see how they affect the skeleton and may be used by forensic anthropologists to assist in identification.

Chapter 5 featured the results of the interviews I conducted over the course of this thesis and provided opinions from practicing forensic anthropologists who provided insight into how this research can be useful in the field.

6.2 Some Ways Forward

Throughout this thesis, the desire for more research was made apparent to me time and time again. Adams et al. (2023, 2024) show the work being done by current forensic anthropologists to open the discussion of transgender identification. In my interviews, participants discussed that this is a question on students' tongues, with students approaching them to supervise projects on the topic as both Dr. Cahn and Dr. Tallman have. Smith, Cahn and Participant 2 mentioned seeing presentations at conferences on gender-diverse identification and calls for anthropologists to be aware of the role they play in the identification of an individual.

Students and anthropologists are not the only ones aware of this problem. The Trans Doe Task Force is organized to combat the long-standing problem of transgender individuals being the victims of violent crimes. Global statistics gathered by Wirtz et al. (2020) found that 2,343 murders of transgender individuals were reported globally between 2008 and 2016. Transgender individuals report being the victims of violent crimes 2.5 times more than cisgender people despite making up only 0.5-1.4% of the population in the United States (Herman, Flores, and O'Neill 2022). Estimates of data are compiled using self-reporting of sex and gender on government census, data from individual studies, and data from police-reported accounts which have been proven to be approximately half the actual experienced violence by transgender individuals (Flores et al. 2021; Gyamerah et al. 2021).

Community engagement emerged throughout the handful of publications and was reiterated by participants, most notably with the desire to follow the idea of "nothing about us without us," ensuring anthropologists do not speak for the transgender community, but rather work alongside them to develop best practices. Flaherty et al. (2023) demonstrated research that aligns with these values, being approached by a transgender community member who was interested in her own identification. This concern is reiterated by the Trans Doe Task Force and its participants who are fearful of how they will be treated in death and utilize the task as advocates for their identity should something happen to them. This is something that Adams et al. (2023, 2024) call on anthropologists to help with; they call upon practitioners to take care with terminology, using the preferred pronouns of an individual once identified, advocating for proper pronoun usage in the field and in reports, and suggesting shifts in terminology to help reflect this.

Shifting terminology within the practice of forensic anthropology is an additional call to action made by Adams et al. (2024). They advocate for the shift to assigned sex at birth, with the secondary idea of a shift to osteological sex, rather than just stating sex on medicolegal forms. When asked about this in interviews, Smith suggested that osteological sex is a more concise way to describe exactly what forensic anthropologists can assess without the concerns voiced by Participant 2, that assigned sex at birth may imply transgender status when it is not necessarily evident. The use of the term osteological sex is the most concise way for anthropologists to relate exactly what they can assess from the skeletal record.

In regard to the identification of transgender individuals, interview participants were asked about both the skeletal record and the role they feel cultural items serve as identifiers. Participants all agreed that the items found at the scene of a crime play an important role, but voiced concerns of over relying on these objects and assigning a cisgender individual as transgender further confusing a case. This is particularly due to the variability of evidence at the scene and the ability of perpetrators to manipulate this evidence. Participants favoured what they could see in the skeletal record for their analysis.

Looking at the role of hormones in bone development, research on the role of puberty blockers and gender-affirming hormone therapy in bone development was perceived to be a nonstarter in identification. When presented with research published on the role of pharmaceutical development, three participants agreed that this is not an area where research should focus. They concluded that the changes would be too minimal to perceive and too hard to attribute specifically to the use of puberty blockers. Smith however saw the value in microscopic bone analysis and desired to conduct research in the future in this area to look for indicators on a microscopic level.

On a macroscopic level, the most promising research found was plastic surgery scarring. The evidence presented in Flaherty et al. (2023) aligns with the preliminary findings presented throughout Chapter 4.2, that evidence of healing surgical scars should be evident in the skeletal record. When presented with the idea that facial feminization surgery or facial masculinization surgery may be evident in the skeletal record and the preliminary findings, interview participants had reservations about continuing this research; not due to the lack of need, but a fear of the implications it could create. All participants warned future researchers to be clear in their writing to be careful to avoid grouping all transgender. Participant 2 was the most enthusiastic about this research stating that more knowledge on how plastic surgery appears in the skeletal record would help identify any individual who has had a procedure and cautioned rather than focus on facial feminization surgery or facial masculinization surgery, future research should be more broadly look at each procedure individually, how it presents and the stages of healing of the

skull. They suggested that this knowledge would be a useful identifying factor in any individual and would benefit identification.

In summary, the conversation of transgender identification is only continuing to grow as transitioning and gender-affirming care becomes more accessible. Initial calls to action for forensic anthropologists have been made to help aid in navigating this shift, but the lack of skeletal effects from transitioning makes the future in this research challenging. Instead, while only some individuals have access to surgical gender affirmation, continuing research in this area would benefit the cisgender community as well. Moving forward with research in this area needs to be done with the consent and help of the transgender community, including community members in the research planning stage as well as in writing academic publications to ensure we avoid speaking for the community to ensure we follow the feedback we have been given of "nothing about us without us" (Charlton 2000; Pfeifer 2022).

Bibliography

- Adams, Donovan M., Samantha H. Blatt, Taylor M. Flaherty, Jaxson D. Haug, Mariyam I. Isa, Amy R. Michael, and Ashley C. Smith. 2023. "Shifting the Forensic Anthropological Paradigm to Incorporate the Transgender and Gender Diverse Community." *Humans* 3 (3): 142–65. https://doi.org/10.3390/humans3030013.
- Adams, Donovan M., Ashley C. Smith, Juliette Bedard, Samantha Blatt, Eman Faisal, Jesse Goliath, Sean Tallman, et al. 2024. "Speaking Truth to Power: Toward a Forensic Anthropology of Advocacy and Activisim." *Humans* 2024 (February):66–90. https://doi.org/10.3390/humans401005.
- Agochukwu-Mmonu, Nnenaya, Asa Radix, Lee Zhao, Danil Makarov, Rachel Bluebond-Langner, A. Mark Fendrick, Elijah Castle, and Carolyn Berry. 2022. "Patient Reported Outcomes in Genital Gender-Affirming Surgery: The Time Is Now." *Journal of PatientReported Outcomes* 6 (1): 39. https://doi.org/10.1186/s41687-022-00446-x.
- Alahmari, Manea M, Ali I AlHilali, Taef A Thabet, Mushabab A Alshahrani, Wejdan A Mobasher, Dalia A Al Mubarak, Abdullah M Alshamrani, Raghad S Gohman, Seham A Alqarni, and Malak M Alqahtani. 2023. "Impact of Medication Adherence on Bone Mineral Density and Fracture Risk in Patients With Osteoporosis: A Systematic Review." *Cureus* 15 (7): e42115. https://doi.org/10.7759/cureus.42115.
- Alfandary, Isabelle. 2019. "Identity between sex and gender in Freud and Lacan." *Journal of the CIPH* 95 (1): 22–43.
- Al-Mamun, Md., Md. Jamal Hossain, Morshed Alam, Md. Shahin Parvez, Bablu Kumar Dhar, and Md. Rabiul Islam. 2022. "Discrimination and Social Exclusion of Third-Gender Population (Hijra) in Bangladesh: A Brief Review." *Heliyon* 8 (10): e10840. https://doi.org/10.1016/j.heliyon.2022.e10840.
- Almstedt Shoepe, Hawley, and Christine M. Snow. 2005. "Oral Contraceptive Use in Young Women Is Associated with Lower Bone Mineral Density than That of Controls." *Osteoporosis International* 16 (12): 1538–44. https://doi.org/10.1007/s00198-005-1868-6.
- Amato, Viola. 2016. "Chapter 6 Screening Intersex on PrimeTime: Intersex in/as a State of Emergency and Popular Culture's Un/Acceptable Interventions." In Intersex Narratives: Shifts in the Representation of Intersex Lives in North American Literature and Popular Culture. Vol. 12. Bielefeld. https://doi.org/10.1515/9783839434192.
- Arron, Joseph R., and Yongwon Choi. 2000. "Bone versus Immune System." *Nature* 408 (6812): 535–36. https://doi.org/10.1038/35046196.
- Atik, Tahir, Özgür Çoğulu, and Ferda Özkınay. 2016. "A Rare Sex Chromosome Aneuploidy: 48,XXYY Syndrome." *Turkish Archives of Pediatrics/Türk Pediatri Arşivi* 51 (2): 106–9. https://doi.org/10.5152/TurkPediatriArs.2016.1551.
- Bachrach, Laura K., and Catherine M. Gordon. 2016. "Bone Densitometry in Children and Adolescents." *Pediatrics* 138 (4): e20162398. https://doi.org/10.1542/peds.2016-2398.

- Balzer, Carsten, and Jan Simon Hutta. 2012. "Transrespect versus Transphobia Worldwide." *Transgender Europe*.
- Barakat, Muna, Raja'a Al-Qudah, Amal Akour, Mona Abu-Asal, Samar Thiab, and Yahya H.
 Dallal Bashi. 2021. "Knowledge and Beliefs about the Use/Abuse of Oral Contraceptive Pills among Males: A Mixed-Method Explanatory Sequential Study in Community Pharmacy Settings." *PLOS ONE* 16 (5): e0251302. https://doi.org/10.1371/journal.pone.0251302.
- Belcastro, Maria Giovanna, Antonio Todero, Gino Fornaciari, and Valentina Mariotti. 2011.
 "Hyperostosis Frontalis Interna (HFI) and Castration: The Case of the Famous Singer Farinelli (1705–1782)." *Journal of Anatomy* 219 (5): 632–37. https://doi.org/10.1111/j.1469-7580.2011.01413.x.
- Black, Michele C., Kathleen C. Basile, Matthew J. Breiding, Sharon G. Smither, Mikel L. Walters, Melissa T. Merrick, Jieru Chen, and Mark R. Stevens. 2011. "National Intimate Partner and Sexual Violence Survey. 2010 Summary Report." Atlanta, Georgia.: National Center for injury Prevention and Control Centers for Disease Control and Prevention. https://www.cdc.gov/violenceprevention/pdf/nisvs_report2010-a.pdf.
- Blau, Soren. 2017. "How Traumatic: A Review of the Role of the Forensic Anthropologist in the Examination and Interpretation of Skeletal Trauma." *Australian Journal of Forensic Sciences* 49 (3): 261–80. https://doi.org/10.1080/00450618.2016.1153715.
- Bourseul, Vincent. 2014. "Gender in Psychoanalysis. The Scope of a Definition." *Recherches en Psychanalyse* 17 (1): 63a–72a. https://doi.org/10.3917/rep.017.0063a.
- Brown, Judith, and WM Roger Louis, eds. 1999. The Oxford History of the British Empire: Volume IV: The Twentieth Century. Oxford University Press. https://doi.org/10.1093/acprof:oso/9780198205647.001.0001.
- Buikstra, J. E., and D. Ubelaker. 1994. *Standards for Data Collection for Human Skeletal Remains*.
- Buikstra, Jane E., ed. 2019. Ortner's Identification of Pathological Conditions in Human Skeletal Remains | ScienceDirect. Third. Elsevier Inc. https://wwwsciencedirectcom.login.ezproxy.library.ualberta.ca/book/9780128097380/ortnersidentification-ofpathological-conditions-in-human-skeletal-remains.
- Bukyya, Jaya Lakshmi, M L. Avinash Tejasvi, Anulekha Avinash, Chanchala H. P., Priyanka Talwade, Mohammed Malik Afroz, Archana Pokala, Praveen Kumar Neela, T K.
 Shyamilee, and Vammi Srisha. 2021. "DNA Profiling in Forensic Science: A Review." *Global Medical Genetics* 8 (4): 135–43. https://doi.org/10.1055/s-0041-1728689.
- Butler, Judith. 2011. *Bodies That Matter: On the Discursive Limits of Sex*. London: Routledge. https://doi.org/10.4324/9780203828274.
- Carpenter, Morgan. 2018. "The 'Normalization' of Intersex Bodies and 'Othering' of Intersex Identities in Australia." *Journal of Bioethical Inquiry* 15 (4): 487–95. https://doi.org/10.1007/s11673-018-9855-8.
- Chaljub, G., R. F. Johnson III, R. F. Johnson Jr., and C. W. Sitton. 1999. "Unusually Exuberant Hyperostosis Frontalis Interna: MRI." *Neuroradiology* 41 (1): 44–45.

https://doi.org/10.1007/s002340050703.

- Charlton, James I. 2000. Nothing About Us Without Us: Disability Oppression and Empowerment. University of California Press.
- Ciancia, Silvia, Vanessa Dubois, and Martine Cools. 2022. "Impact of Gender-Affirming Treatment on Bone Health in Transgender and Gender Diverse Youth." *Endocrine Connections* 11 (11): e220280. https://doi.org/10.1530/EC-22-0280.
- Coleman, E., Radix, A. E., Bouman, W. P., Brown, G. R., de Vries, A. L. C., Deutsch, M. B., ... Arcelus, J. 2022. "Standards of Care for the Health of Transgender and Gender Diverse People, Version 8." *International Journal of Transgender Health* 23(sup1): S1–S259. https://doi.org/10.1080/26895269.2022.2100644.
- Crabtree, Nicola J, Nicholas J Shaw, Nicholas J Bishop, Judith E Adams, M Zulf Mughal, Paul Arundel, Mary S Fewtrell, et al. 2017. "Amalgamated Reference Data for Size-Adjusted Bone Densitometry Measurements in 3598 Children and Young Adults—the ALPHABET Study." *Journal of Bone and Mineral Research* 32 (1): 172–80. https://doi.org/10.1002/jbmr.2935.
- Creighton, Sarah. 2001. "Surgery for Intersex." *Journal of the Royal Society of Medicine* 94 (5): 218–20.
- Daniels, Kimberly, and Joyce C. Abma. 2022. "Products Data Briefs Number 327 December 2018." CDC. October 31, 2022. https://www.cdc.gov/nchs/products/databriefs/db327.htm.
- Dar, Hamid Y., Zaffar Azam, Rajaneesh Anupam, Rajesh K. Mondal, and Rupesh K. Srivastava. 2018. "Osteoimmunology: The Nexus between Bone and Immune System." *Frontiers in Bioscience*, 464–92.
- Dash, Hiram R., Neha Rawat, and Surajit Das. 2019. "Alternatives to Amelogenin Markers for Sex Determination in Humans and their Forensic Relevance." *Molecular Biology Reports*. 47: 2347-2360.
- Davies, Charlie. 2022. "Flowers of Pride: The Green Carnation." Walledgardenatmells. June 24, 2022. https://www.thewalledgardenatmells.co.uk/post/flowers-of-pride-the-greencarnation.
- De Angelis, Danilo, and Cristina Cattaneo. 2015. "Implant Bone Integration Importance in Forensic Identification." *Journal of Forensic Sciences* 60 (2): 505–8. https://doi.org/10.1111/1556-4029.12640.
- DeCherney, Alan. 1996. "Bone-Sparing Properties of Oral Contraceptives." American Journal of Obstetrics and Gynecology 174 (1): 15–20. https://doi.org/10.1016/S00029378(96)70366-6.
- Deschamps-Baly, Jordan, Caitlin L. Sacher, Jennifer Fick, and Douglas Ousterhout. 2017. "First Female-to-Male Facial Confirmation Surgery with Description of a New Procedure for Masculinization of the Thyroid Cartilage (Adam's Apple)" 139 (4): 883e–87.
- DesMarais, Angel, Zuzana Obertova, and Daniel Franklin. 2024. "The Influence of Age on Greater Sciatic Notch Morphology: Testing the Walker Method in an Australian

Population." *International Journal of Legal Medicine* 138 (1): 239–47. https://doi.org/10.1007/s00414-023-02988-1.

- D'hoore, Laurens, and Guy T'Sjoen. 2022. "Gender-Affirming Hormone Therapy: An Updated Literature Review with an Eye on the Future." *Journal of Internal Medicine* 291 (5): 574–92. https://doi.org/10.1111/joim.13441.
- Di Meglio, Giuseppina, and Elizabeth Yorke. 2019. "Universal Access to No-Cost Contraception for Youth in Canada | Canadian Paediatric Society." *Canadian Paediatric Society* (blog). https://cps.ca/documents/position/universal-access-to-no-cost-contraception-for-youthincanada.
- Dickens, Bernard M. 2018. "Management of Intersex Newborns: Legal and Ethical Developments." *International Journal of Gynecology & Obstetrics* 143 (2): 255–59. https://doi.org/10.1002/ijgo.12573.
- Editorial Staff, QueerEvents. 2018. "Colour Codes: Flagging in the Queer Community." *Queer Events* (blog). February 14, 2018. https://www.queerevents.ca/queercorner/blog/history/queers-community-flagging.
- Eng, Jacqueline, Quanchao Zhang, and Hong Zhu. 2010. "Skeletal Effects of Castration on Two Eunuchs of Ming China." *Anthropological Science* 118 (August):107–16. https://doi.org/10.1537/ase.090430.
- Evans, Michelle D., and Hillary J. Knepper. 2020. "Building Inclusive PA Classrooms: The Diversity Inclusion Model." *Public Administration Comittee* 39 (1). https://journalssagepubcom.login.ezproxy.library.ualberta.ca/doi/10.1177/0144739420937762.
- Fausto-Sterling, Anne. 1993. "The Five Sexes: Why Male and Female Are Not Enough." *The Sciences* 33 (2): 20–25.
- Fausto-Sterling, Anne. 2000. "The Five Sexes, Revisted." *The Sciences* 40 (4): 18–23. https://doi.org/10.1002/j.2326-1951.2000.tb03504.x.
- Feder, Ellen K., and Katrina Karkazis. 2008. "What's in a Name?: The Controversy over 'Disorders of Sex Development." *Hastings Center Report* 38 (5): 33–36. https://doi.org/10.1353/hcr.0.0062.
- Ferguson, Sian. 2020. "How Cisgender and Transgender Guys Use Hormonal Birth Control." Healthline. 2020. https://www.healthline.com/health/what-happens-if-a-guy-takesbirthcontrol#effects-vary.
- Flaherty, Taylor M., Jennifer F. Byrnes, and Antonella Maddalena. 2023. "Misgendering a Transgender Woman Using FORDISC 3.1: A Case Study." *Forensic Science International: Synergy* 7 (June):100342. https://doi.org/10.1016/j.fsisyn.2023.100342.
- Flaherty, Taylor M., Liam J. Johnson, Katharine C. Woollen, Dayanira Lopez, Katherine Gaddis, SaMoura L. Horsley, and Jennifer F. Byrnes. 2023. "Speaking of Sex: Critical Reflections for Forensic Anthropologists." *Humans* 3 (4): 251–70. https://doi.org/10.3390/humans3040020.

- Flores, Andrew R., Ilan H. Meyer, Lynn Langton, and Jody L. Herman. 2021. "Gender Identity Disparities in Criminal Victimization: National Crime Victimization Survey, 2017-2018." *American Journal of Public Health* 111 (4): 726–29. https://doi.org/10.2105/AJPH.2020.306099.
- Freud, Sigmund. 1895. The Standard Edition of the Complete Psychological Works of Sigmund Freud, Volume II (1893-1895): Studies on Hysteria. https://pepweb.org/browse/document/SE.002.0000A?page=PR0004.
- Gamble, Julia A., Victor Spicer, Mercedes Hunter, Yang Lao, Robert D. Hoppy, Dorthe Dangvard Pedersen, John A. Wilkins, René P. Zahedi. 2024. "Advancing Sex Estimation from Amelogenin: applications to Archaeological, Deciduous, and Fragmentary Dental Enamel." *Journal of Archaeological Science: Reports* 54: 1-9.

Garnero, P., E. Sornay-Rendu, and P. D. Delmas. 1995. "Decreased Bone Turnover in Oral Contraceptive Users." *Bone* 16 (5): 499–503. https://doi.org/10.1016/87563282(95)00075-O.

- Garofalo, Evan M., and Heather M. Garvin. 2020. "Chapter 4 The Confusion between Biological Sex and Gender and Potential Implications of Misinterpretations." In Sex Estimation of the Human Skeleton, edited by Alexandra R. Klales, 35–52. Academic Press. https://doi.org/10.1016/B978-0-12-815767-1.00004-3.
- Ghorayshi, Azeen. 2023. "England Limits Use of Puberty-Blocking Drugs to Research Only." *The New York Times*, June 9, 2023, sec. Health. https://www.nytimes.com/2023/06/09/health/puberty-blockers-transgenderchildrenbritain-nhs.html.
- Gill, P., A. J. Jeffreys, and D. J. Werrett. 1985. "Forensic Application of DNA 'Fingerprints."" *Nature* 318 (6046): 577–79. https://doi.org/10.1038/318577a0.
- Goldschmidt, Richard. 1917. "Intersexuality and the Endocrine Aspect of Sex." *Endocrinology* 1 (4): 433–56.
- Goodman, Leo A. 1961. "Snowball Sampling." Annals of Mathematical Statistics 32:148-70.
- Goymann, Wolfgang, Henrik Brumm, and Peter M. Kappeler. 2023. "Biological Sex Is Binary, Even Though There Is a Rainbow of Sex Roles." *BioEssays* 45 (2): 2200173. https://doi.org/10.1002/bies.202200173.
- Gyamerah, Akua O., Glenda Baguso, Edda Santiago-Rodriguez, Aria Sa'id, Sean Arayasirikul, Jess Lin, Caitlin M. Turner, et al. 2021. "Experiences and Factors Associated with Transphobic Hate Crimes among Transgender Women in the San Francisco Bay Area: Comparisons across Race." *BMC Public Health* 21 (1): 1053. https://doi.org/10.1186/s12889-021-11107-x.
- Hale, Amanda R., and Ann H. Ross. 2018. "Scanning Skeletal Remains for Bone Mineral Density in Forensic Contexts." *Journal of Visualized Experiments: JoVE*, no. 131 (January), 56713. https://doi.org/10.3791/56713.
- Halkitis, Perry N. 2019. "The Stonewall Riots, the AIDS Epidemic, and the Public's Health." *A J Public Health* 109 (6): 851-852. https://doi.org/10.2105%2FAJPH.2019.305079

- Hann, Michael D., and Tae-Geon Kwon. 2022. "Zygoma and Mandibular Angle Reduction -ClinicalKey." Oral and Maxillofacial Surgery Clinics 35 (1): 83–96.
- Haraldsen I. R., Haug E., Falch J., Egeland T., Opjordsmoen S. 2007. "Cross-Sex Pattern of Bone Mineral Density in Early Onset Gender Identity Disorder." *Horm Behav.* 52(3):334-43. doi: 10.1016/j.yhbeh.2007.05.012. Epub 2007 Jun 2. PMID: 17604029.
- Herman, Jody L, Andrew R Flores, and Kathryn K O'Neill. 2022. "How Many Adults and Youth Identify as Transgender in the United States?"
- Hershkovitz, I., C. Greenwald, B. M. Rothschild, B. Latimer, O. Dutour, L. M. Jellema, and S. Wish-Baratz. 1999. "Hyperostosis Frontalis Interna: An Anthropological Perspective." *American Journal of Physical Anthropology* 109 (3): 303–25. https://doi.org/10.1002/(SICI)1096-8644(199907)109:3<303::AID-AJPA3>3.0.CO;2-I.
- Hinchy, Jessica. 2019. Governing Gender and Sexuality in Colonial India: The Hijra, c.1850– 1900. 1st ed. Cambridge University Press. https://doi.org/10.1017/9781108592208.
- Houghton, Philip. 1974. "The Relationship of the Pre-auricular Groove of the Llium to Pregnancy." American Journal of Physical Anthropology 41 (3). https://doiorg.login.ezproxy.library.ualberta.ca/10.1002/ajpa.1330410305.
- Hoyt, Alia, and Karen Kirkpatrick. 2022. "What If a Man Takes Birth Control Pills?" *Howstuffworks* (blog). https://science.howstuffworks.com/science-vs-myth/whatif/whatif-man-takes-birth-control-pills.htm.
- Idrus, Nurul Ilmi, and Takeo David Hymans. 2014. "Balancing Benefits and Harm: Chemical Use and Bodily Transformation among Indonesia's Transgender Waria." *The International Journal on Drug Policy* 25 (4). https://doi.org/10.1016/j.drugpo.2014.06.012.
- Lahood, Grant. 2013. Intersexion. Auckland, N.Z.: Ponsonby Productions.
- Lal, Vinay. 1999. "Not This, Not That: The Hijras of India and the Cultual Politics of Sexuality" Out Front: Lesbians, Gays, and the Struggle for Workplace Rights 61: 119-140
- Lopez LM, Grimes DA, Schulz KF, Curtis KM., and Chen M. 2014. "Steroidal contraceptives: effect on bone fractures in women." Cochrane Database of Systematic Reviews 6. (CD006033). DOI: 10.1002/14651858.CD006033.pub5
- James, Sandy E., Jody L. Herman, Susan Rankin, Mara Keisling, Lisa Mottet, and Ma'ayan Anafi. 2016. "The Report of the 2015 U.S. Transgender Survey." Washington, BC: *National Center for Transgender Equality.*
- Jefferson, Kevin, Torsten B. Neilands, and Jae Sevelius. 2013. "Transgender Women of Color: Discrimination and Depression Symptoms." *Ethnicity and Inequalities in Health and Social Care* 6 (4): 121–36. https://doi.org/10.1108/EIHSC-08-2013-0013.
- Jodarski, Colleen, Rylee Duncan, Erin Torres, Rachel Gore, Armin Raznahan, and Morgan Similuk. 2023. "Understanding the Phenotypic Spectrum and Family Experiences of XYY Syndrome: Important Considerations for Genetic Counseling." *Journal of Community Genetics* 14 (1): 17–25. https://doi.org/10.1007/s12687-022-00630-y.

- Johnson, Timothy P. 2014. "Snowball Sampling: Introduction." *Wiley StatsRef: Encyclopedia of Biostatistics*, September. https://onlinelibrarywileycom.login.ezproxy.library.ualberta.ca/doi/10.1002/9781118445112.stat05720.
- Joseph, Michael. 2008. "Endodontic Treatment in Three Taurodontic Teeth Associated with 48,XXXY Klinefelter Syndrome: A Review and Case Report." Oral Surgery, Oral Medicine, Oral Pathology, Oral Radiology, and Endodontology 105 (5): 670–77. https://doi.org/10.1016/j.tripleo.2007.11.015.
- Joseph, Tobin, Joanna Ting, and Gary Butler. 2019. "The Effect of GnRH Analogue Treatment on Bone Mineral Density in Young Adolescents with Gender Dysphoria: Findings from a Large National Cohort." *Journal of Pediatric Endocrinology and Metabolism* 32 (10): 1077–81. https://doi.org/10.1515/jpem-2019-0046.
- Karsten, Jordan K. 2018. "A Test of the Preauricular Sulcus as an Indicator of Sex." *American Journal of Physical Anthropology* 165 (3): 604–8. https://doi.org/10.1002/ajpa.23372.
- Kelley, Samantha R., and Sean D. Tallman. 2022. "Forensic Sciences | Free Full-Text | Population-Inclusive Assigned-Sex-at-Birth Estimation from Skull Computed Tomography Scans." *Forensic Sciences* 2:321–48.
- Keyhan, Seied, Seifollah Hemmat, Peiman Mehriar, and Arash Khojasteh. 2015. "Advanced Adjunct Orthosurgical Esthetic Procedures." In https://doi.org/10.5772/59088.
- Khartade, Harshwardhan K., and Vikas Meshram. 2020. "Identification of Skeletal Remains by Orthopaedic Implant: A Case Report and Brief Review of the Literature - Harshwardhan K Khartade, Vikas Meshram, Shashidhar P Garg, Dhruwa K Mishra, 2021." *Dhurwa K*. 61 (2). https://doi-org.login.ezproxy.library.ualberta.ca/10.1177/0025802420975304.
- Kilmer, Katie, and Heather Garvin. 2020. "Outline Analysis of Sex and Population Variation in Greater Sciatic Notch and Obturator Foramen Morphology with Implications for Sex Estimation." *Forensic Science International* 314: 110346. https://doi.org/10.1016/j.forsciint.2020.110346.
- Klales, Alexandra R., ed. 2020. Sex Estimation of the Human Skeleton: History Methods and *Emerging*.
- Klales, Alexandra R., Stephen D. Ousley, and Jennifer M. Vollner. 2012. "A Revised Method of Sexing the Human Innominate Using Phenice's Nonmetric Traits and Statistical Methods." *American Journal of Physical Anthropology* 149 (1): 104–14. https://doi.org/10.1002/ajpa.22102.
- Klink, Daniel, Martine Caris, Annemieke Heijboer, Michael van Trotsenburg, and Joost Rotteveel. 2015. "Bone Mass in Young Adulthood Following Gonadotropin-Releasing Hormone Analog Treatment and Cross-Sex Hormone Treatment in Adolescents With Gender Dysphoria." *The Journal of Clinical Endocrinology & Metabolism* 100 (2): E270–75. https://doi.org/10.1210/jc.2014-2439.
- Krishnappa, Pramod, Esaú Fernández-Pascual, and Juan Ignacio Martinez-Salamanca. 2022.

"Patient and Device Selection." In *Penile Implant Surgery: Contemporary Challenges and Controversies*, edited by Eduardo P. Miranda and John P. Mulhall, 1–16. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-82363-4 1.

- Lal, Vinay. 1999. "Not This, Not That: The Hijras of India and the Cultural Politics of Sexuality." *Social Text*, no. 61, 119–40.
- Lam, Jason C., and Taif Mukhdomi. 2023. "Kyphosis." *StatPearls [Internet]*. Treasure Island (FL): StatPearls Publishing. https://www.ncbi.nlm.nih.gov/books/NBK558945/.
- Lanzarotta, Adam, and Caroline Machal Kelley. 2016. "Forensic Analysis of Human Autopsy Tissue for the Presence of Polydimethylsiloxane (Silicone) and Volatile Cyclic Siloxanes Using Macro FT-IR, FT-IR Spectroscopic Imaging and Headspace GC-MS." *Journal of Forensic Sciences* 61 (3): 867–74. https://doi.org/10.1111/1556-4029.13018.
- Latham, Krista E., and Jessica J. Miller. 2018. "DNA Recovery and Analysis from Skeletal Material in Modern Forensic Contexts." *Forensic Sciences Research* 4 (1): 51–59. https://doi.org/10.1080/20961790.2018.1515594.
- Lloyd, Tom, James R. Buchanan, Gregory R. Ursino, Cathleen Myers, Gerald Woodward, and David R. Halbert. 1989. "Long-Term Oral Contraceptive Use Does Not Affect Trabecular Bone Density - ScienceDirect." https://wwwsciencedirectcom.login.ezproxy.library.ualberta.ca/science/article/abs/pii/0002937889904 572.
- Marchini, Antonio, Tsutomu Ogata, and Gudrun A. Rappold. 2016. "A Track Record on SHOX: From Basic Research to Complex Models and Therapy." *Endocrine Reviews* 37 (4): 417–48. https://doi.org/10.1210/er.2016-1036.
- Meyer, Doug. 2015. Violence against Queer People: Race, Class, Gender, and the Persistence of Anti-LGBT Discrimination. Violence against Queer People: Race, Class, Gender, and the Persistence of Anti-LGBT Discrimination. Piscataway, NJ, US: Rutgers University Press.
- Mitchell-Brody, Maryse, and Andrea J. Ritchie. 2010. "Hate Violence against the Lesbian, Gay, Bisexual, Transgender and Queer Communities in the United States in 2009." National Coalition of Anti-Violence Programs. https://avp.org/wpcontent/uploads/2017/04/2009 NCAVP HV Report.pdf.
- Mohamad, Nur-Vaizura, Ima-Nirwana Soelaiman, and Kok-Yong Chin. 2016. "A Concise Review of Testosterone and Bone Health." *Clinical Interventions in Aging* 11: 1317–24. https://doi.org/10.2147/CIA.S115472.
- Moilanen, Ulla, Tuija Kirkinen, Nelli-Johanna Saari, Adam B. Rohrlach, Johannes Krause,
 Päivi Onkamo and Eliana Salmela. 2022. "A Woman with a Sword? Weapon Grave at
 Suontaka Vesitorninmäki, Finland." *European Journal of Archaeology* 25 (1): 42-60.
- Moore, Megan K. 2013. "Sex Estimation and Assessment." In *Reasearch Methods in Human Skeletal Biology*.
- Murphy, Ryan, and Brad Falchuk, dirs. 2013. "Sweet Dreams." Glee.
- Nanda, Serena. 1986. "The Hijras of India:" *Journal of Homosexuality* 11 (3–4): 35–54. https://doi.org/10.1300/J082v11n03_03.

- Nos, Andrea L., David A. Klein, Terry A. Adirim, Natasha A. Schvey, Elizabeth Hisle-Gorman, Apryl Susi, and Christina M. Roberts. 2022. "Association of Gonadotropin-Releasing Hormone Analogue Use With Subsequent Use of Gender-Affirming Hormones Among Transgender Adolescents." *JAMA Network Open* 5 (11): e2239758. https://doi.org/10.1001/jamanetworkopen.2022.39758.
- Ousterhout, Douglas. 1987. "Feminization-of-the-Forehead-1987-Ousterhout.Pdf." *Plastic and Reconstructive Surgery* 79 (5): 701–11.
- Ousterhout, Douglas. 2010. Facial Feminiztion Surgery: A Guide for the Transgendered Woman. Addicus Books.
- Ousterhout, Douglas. 2011. "Dr. Paul Tessier and Facial Skeletal Masculinization." *Annals of Plastic Surgery* 67 (6): S10. https://doi.org/10.1097/SAP.0b013e31821835cb.
- Panday, Keshav, Amitha Gona, and Mary Beth Humphrey. 2014. "Medication-Induced Osteoporosis: Screening and Treatment Strategies." *Therapeutic Advances in Musculoskeletal Disease* 6 (5): 185–202. https://doi.org/10.1177/1759720X14546350.
- Peitzmeier, Sarah M., Mannat Malik, Shanna K. Kattari, Elliot Marrow, Rob Stephenson, Madina Agénor, and Sari L. Reisner. 2020. "Intimate Partner Violence in Transgender Populations: Systematic Review and Meta-Analysis of Prevalence and Correlates." *American Journal of Public Health* 110 (9): e1–14. https://doi.org/10.2105/AJPH.2020.305774.
- Peoples, Landon. 2019. "The History Of The Handkerchief Code LGBTQ Language." *Refinery* 29 (blog). June 12, 2019. https://www.refinery29.com/en-us/2018/05/200229/lgbtqsecrethandkerchief-code-language.
- Pfeifer, Whitney. 2022. "From 'Nothing About Us Without Us' to 'Nothing Without Us." Text. March 2022. https://www.ndi.org/our-stories/nothing-about-us-without-usnothingwithout-us.
- Phenice, T. W. 1969. "A Newly Developed Visual Method of Sexing the Os Pubis." American Journal of Physical Anthropology 30 (2): 297–301. https://doi.org/10.1002/ajpa.1330300214.
- Porter, Andrew, and WM Roger Louis, eds. 1999. The Oxford History of the British Empire: Volume III: The Nineteenth Century | Oxford Academic. Vol. 3. Oxford University Press. https://academic.oup.com/book/1389.
- Rajabzadeh, Mahboubeh, Nafiseh Taheri, and Omid Jazayeri. 2022. "49,XXXXY Syndrome: A Case Study and a Systematic Review of Clinical Features among the Iranian Population." *Clinical Case Reports* 10 (9): e6342. https://doi.org/10.1002/ccr3.6342.
- Ramachandran, K. 2008. "Breast Augmentation." *Indian Journal of Plastic Surgery : Official Publication of the Association of Plastic Surgeons of India* 41 (Suppl): S41–47.
- Ramchand, Sabashini K, Natalie L David, Benjamin Z Leder, and Joy N Tsai. 2019. "Bone Mineral Density Response With Denosumab in Combination With Standard or HighDose Teriparatide: The DATA-HD RCT." *The Journal of Clinical Endocrinology and Metabolism* 105 (3): 890–97. https://doi.org/10.1210/clinem/dgz163.

- Randolph, John F. Jr., and Karvonen-Gutierrez. 2021. "Chapter 23: Reproductive and Hormonal Factors and the Risk for Osteoporosis." In *Marcus and Feldman's Osteoporosis (Fifth Edition)* Academic Press:545-573.
- Rehmann-Sutter, Christoph, Olaf Hiort, Ulrike M. Krämer, Lisa Malich, and Malte Spielmann. 2023. "Is Sex Still Binary?" *Medizinische Genetik* 35 (3): 173–80. https://doi.org/10.1515/medgen-2023-2039.
- Reid, MarySarah Grace. 2018. "What Is the M/F Problem?': Critical Discourse Analysis in Anne Fausto-Sterling's Sexing the Body: Gender Politics and the Construction of Sexuality."
- Reusch, K. 2013. "Chapter 6: Results and Initial Conclusions." University of Oxford. https://ora.ox.ac.uk/objects/uuid:b8118fe7-67cb-4610-9823b0242dfe900a/download_file?file_format=application%2Fpdf&safe_filename=THE SIS0 6&type_of_work=Thesis.
- Ricker, Emily A., Koltun, Kristen J., and de la Motte, Sarah J., 2024. "Hormonal Contraception and Medical Readiness for Female Service Members." *American Journal of Obstetrics* and Gynecology. https://doi.org/10.1016/j.ajog.2024.05.044.
- Ridgeway, Beri M., Beatriz E. Arias, and Matthew D. Barber. 2008. "Variation of the Obturator Foramen and Pubic Arch of the Female Bony Pelvis." *American Journal of Obstetrics* and Gynecology 198 (5): 546.e1-546.e4. https://doi.org/10.1016/j.ajog.2008.01.055.
- Riggs BL, Khosla S, Melton LJ 3rd. 1998. "A Unitary Model for Involutional Osteoporosis: Estrogen Deficiency Causes Both Type I and type II Osteoporosis in Postmenopausal Women and Contributes to Bone Loss in Aging Men. *J Bone Miner Res* 13(5):763-73. doi: 10.1359/jbmr.1998.13.5.763. PMID: 9610739.
- Roberts, Alissa J., and Patricia Y. Fechner. 2020. "Description of Turner Syndrome." In Turner Syndrome: Pathophysiology, Diagnosis and Treatment, edited by Patricia Y. Fechner, 1– 12. Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-341503_1.
- Roca-Rada, Xavier, Sofia Tereso, Adam B. Rohrlach, André Brito, Matthew P. Williams, Cláudia Umbelino, Francisco Curate, Ira W Deveson, Yassine Souilmi, Antónia Amorim, Pedro C. Carvalho, Bastien Llamas, João C. Teixeira. 2020. "The 1000-year-old case of Klinefelter's Syndrome Diagnosed by Integrating Morphology, Osteology and Genetics." *Clinical Picture* 400: 691-692
- Rogers, Laura M., Siân E. Halcrow, Torsten Kleffmann, and Charlotte L. King. 2024.
 "Comparing Walker's (2008) Skull Trait Sex Estimation Standard to Proteomic Sex Estimation for a Group of South Asian Individuals." *Forensic Science International: Synergy* 8: 100450. https://doi.org/10.1016/j.fsisyn.2023.100450.
- Reusch, K. 2013. "That Which Was Missing': the Archaeology of Castration." PhD thesis, Oxford University, UK.
- Sax, Leonard. 2002. "How Common Is Intersex? A Response to Anne Fausto-Sterling." *Journal* of Sex Research 39 (3): 174–78. https://doi.org/10.1080/00224490209552139.

- Schall, Jenna L., Tracy L. Rogers, and Jordan C. Deschamps-Braly. 2020. "Breaking the Binary: The Identification of Trans-Women in Forensic Anthropology." *Forensic Science International* 309 (April):110220. https://doi.org/10.1016/j.forsciint.2020.110220.
- Sheu, Angela, and Terry Diamond. 2016. "Bone Mineral Density: Testing for Osteoporosis." *Australian Prescriber* 39 (2): 35–39. https://doi.org/10.18773/austprescr.2016.020.
- Simpson, Ellie K., Ross A. James, David A. Eitzen, and Roger W. Byard. 2007. "Role of Orthopedic Implants and Bone Morphology in the Identification of Human Remains | Office of Justice Programs," March. https://www.ojp.gov/ncjrs/virtuallibrary/abstracts/role-orthopedic-implants-and-bonemorphology-identification-human.
- Skoglund, Pontus, Jan Storå, Anders Götherström, Mattias Jakobsson. 2013. "Accurate Sex Identification of Ancient Human Remains using DNA shotgun Sequencing." *Journal of Archaeological Science* 40: 4477-4482.
- Soon, Judith A., Anne Marie Whelan, Nesé Yuksel, and Sally Rafie. 2021. "Enhancing Access to Contraception through Pharmacist Prescribing across Canada." *Canadian Pharmacists Journal* : *CPJ* 154 (6): 356–62. https://doi.org/10.1177/17151635211034534.
- Sözen, Tümay, Lale Özışık, and Nursel Çalık Başaran. 2016. "An Overview and Management of Osteoporosis." *Eur J Rheumatol* 4 (1): 46–56.
- Specker, Bonny L., and Eckhard Schoenau. 2005. "Quantitative Bone Analysis in Children: Current Methods and Recommendations." *The Journal of Pediatrics* 146 (6): 726–31. https://doi.org/10.1016/j.jpeds.2005.02.002.
- Surrey E. S. 2023. "GnRH agonists in the treatment of symptomatic endometriosis: a review." *F&S Reports.* 4(2):40-45. doi: 10.1016/j.xfre.2022.11.009.
- Takayanagi, Hiroshi. 2009. "Osteoimmunology and the Effects of the Immune System on Bone." *Nature Reviews Rheumatology* 5 (12): 667–76. https://doi.org/10.1038/nrrheum.2009.217.
- Tallman, Sean D., Caroline D. Kincer, and Eric D. Plemons. 2022. "Centering Transgender Individuals in Forensic Anthropology and Expanding Binary Sex Estimation in Casework and Research." *Forensic Anthropology* 5 (2): 161–80. https://doi.org/10.5744/fa.2020.0030.
- Tandler, J., and S. Grosz. 1909. *Uber Den Einfluss Der Kastraten Auf Den Organismus I. Beschreibung Eines Eunuchen Skeletes*. Vol. 27. Arch. Entwick Mech. Org.
- Tandler, J., and S. Grosz. 1910. Uber Den Einfluss Der Kastraten Auf Den Organismus II. Die Skopzen. Vol. 30. Rch. Entwick Mech. Org.
- Thomas, Richard M. 2020. "Chapter 21 Sex Determination Using DNA and Its Impact on Biological Anthropology." In Sex Estimation of the Human Skeleton, edited by Alexandra R. Klales, 343–50. Academic Press. https://doi.org/10.1016/B978-0-12-815767-1.00021-3.
- Trans Doe. 2024. "Trans Doe Task Force Inc." January 2024. https://transdoetaskforce.org/.

- Truman, Jennifer L., and Rachel E. Morgan. 2022. "Violent Victimization by Sexual Orientation and Gender Identity, 2017–2020 | Bureau of Justice Statistics." NCJ Number 304277. https://bjs.ojp.gov/library/publications/violent-victimization-sexual-orientationandgender-identity-2017-2020.
- Turingan, R.S., J. Brown, and L. Kaplun. et al. 2020. "Identification of human remains using Rapid DNA analysis." Int J Legal Med 134: 863–872. https://doi.org/10.1007/s00414019-02186-y.
- Twohey, Megan, and Christina Jewett. 2022. "They Paused Puberty, but Is There a Cost?" The New York Times, November 14, 2022, sec. Health. https://www.nytimes.com/2022/11/14/health/puberty-blockers-transgender.html.
- Ubelaker, Douglas H. 2018. "Recent Advances in Forensic Anthropology." *Forensic Sciences Research* 3 (4): 275–77. https://doi.org/10.1080/20961790.2018.1466384.
- Van Kesteren, P., Lips, P., Gooren, L.J.G., Asscheman, H. and Megens, J. 1998. "Long-Term Follow-up of Bone Mineral Density and Bone Metabolism in Transsexuals Treated with Cross-Sex Hormones." *Clinical Endocrinology* 48: 347-354. https://doi.org/10.1046/j.1365-2265.1998.00396.x
- Vincent, Ben. 2020. Non-Binary Genders: Navigating Communities, Identities, and Healthcare. 1st ed. Bristol University Press. https://doi.org/10.2307/j.ctv138wrbg.
- Vlot, Mariska C., Daniel T. Klink, Martin den Heijer, Marinus A. Blankenstein, Joost Rotteveel, Annemieke C. Heijboer. 2017. "Effect of pubertal suppression and cross-sex hormone therapy on bone turnover markers and bone mineral apparent density (BMAD) in transgender adolescents." *Bone* 95: 11-19. https://doi.org/10.1016/j.bone.2016.11.008.
- Wang, Qingju, Markku Alén, Patrick H. F. Nicholson, Jussi M. Halleen, Sari L. Alatalo, Claes Ohlsson, Harri Suominen, and Sulin Chang. 2006."Differential Effects of Sex Hormones on Peri- and Endocortical Bone Surfaces in Pubertal Girls." *The Journal of Clinical Endocrinology and Metabolism* 91 (1): 277-282.
- Wattendorf, Daniel J., and Maximilian Muenke. 2005. "Klinefelter Syndrome." *American Family Physician* 72 (11): 2259–62.
- Wilson, Jean D., and Claus Roehrborn. 1999. "Long-Term Consequences of Castration in Men: Lessons from the Skoptzy and the Eunuchs of the Chinese and Ottoman Courts." *The Journal of Clinical Endocrinology & Metabolism* 84 (12): 4324–31. https://doi.org/10.1210/jcem.84.12.6206.
- Wirtz, Andrea L., Tonia C. Poteat, Mannat Malik, and Nancy Glass. 2020. "Gender-Based Violence Against Transgender People in the United States: A Call for Research and Programming." *Trauma, Violence, & Abuse* 21 (2): 227–41. https://doi.org/10.1177/1524838018757749.
- Yang, Wen, Mingquan Zhou, Pengfei Zhang, Guohua Geng, Xiaoning Liu, and Haibo Zhang. 2020. "Skull Sex Estimation Based on Wavelet Transform and Fourier Transform." *BioMed Research International*: 8608209. https://doi.org/10.1155/2020/8608209.

Yaşar, Berkay, and Mehmet Sağır. 2023. "Testing of Morphological Sex Estimation Traits with a Sex-Known Collection: Ottoman Period Skulls." *International Journal of Osteoarchaeology* 33 (6): 1042–51. https://doi.org/10.1002/oa.3265.

Yu, Jessica, dir. 2006. "Begin the Begin." Greys Anatomy.

Yun, Yeoul, Dongyoung Kim, and Eun Sil Lee. 2021. "Effect of Cross-Sex Hormones on Body Composition, Bone Mineral Density, and Muscle Strength in Trans Women." *Journal of Bone Metabolism* 28 (1): 59–66. https://doi.org/10.11005/jbm.2021.28.1.59.

Zanatta, Alberto, Fabio Zampieri, Giuliano Scattolin, and Maurizio Rippa Bonati. 2016.

"Occupational Markers and Pathology of the Castrato Singer Gaspare Pacchierotti (1740–1821)." *Scientific Reports* 6 (1): 28463. https://doi.org/10.1038/srep28463.

Appendices

Appendix A – Ethics Approval



RESEARCH ETHICS OFFICE 2-01 North Power Plant (NPP) 11312 - 89 Ave NW Edmonton, Alberta, Canada T6G 2N2 Tet: 760.492.0459 www.uab.ca/reo

Notification of Approval

Date:	Septebmer 25, 2023
Study ID:	Pro00133307
Principal Investigator:	Brianne Bertram
Study Supervisor:	Sandra Garvie-Lok
Study Title:	Beyond Binary Identification: New ideas of gender and how they fit into forensic anthropology
Approval Expiry Date:	September 24, 2024
Sponsor/Funding Agency:	SSHRC - Social Sciences and Humanities Research Council

Sponson running Agency.

Thank you for submitting the above study to the Research Ethics Board 1. Your application has been reviewed and approved on behalf of the committee.

Approved Documents:

Letter of Initial Contact
Updated Letter of Initial Contact
Consent Forms
Updated Consent Form Version 2
Questionnaires, Cover Letters, Surveys, Tests, Interview Scripts, etc.
Interview Guide Re-Upload

Any proposed changes to the study must be submitted to the REB for approval prior to implementation. A renewal report must be submitted next year prior to the expiry of this approval if your study still requires ethics approval. If you do not renew on or before the renewal expiry date, you will have to re-submit an ethics application.

Approval by the REB does not constitute authorization to initiate the conduct of this research. The Principal Investigator is responsible for ensuring required approvals from other involved organizations (e.g. universities/colleges, community organizations, school boards) are obtained, before the research begins.

Sincerely,

Dr. Theresa Garvin, PhD, MUA, BA Chair, Research Ethics Board 1

Note: This correspondence includes an electronic signature (validation and approval via an online system).

SSHRC



FACULTY OF ARTS DEPARTMENT OF ANTHROPOLOGY

Letter of Initial Contact

[DATE]

[INTERVIEWEE NAME]

Subject: Requesting permission to conduct an interview

Dear Dr. [LAST NAME]

I hope this email finds you well. I am Brianne Bertram, a master's student from the University of Alberta in the Department of Anthropology. I am contacting you to inquire about scheduling an interview for my master's thesis. My research revolves around looking at gender in forensic anthropology. More specifically, I'm interested in how we can potentially identify transgender individuals from their skeletal remains. I am reaching out to you because I have noted from your academic publications and presentations that you are also interested in this topic.

My research study is titled *Beyond Binary Identification: New ideas of gender and how they fit into forensic anthropology* (Pro00133307). My objectives are to discuss current ideas for identifying transgender individuals, the existing literature on this topic, how you would report on potentially transgender remains in a case report, and where you believe research in this area should go next.

This interview will be audio recorded and transcribed for the purpose of my master's thesis, but you will have the choice to be completely anonymized in your answers if you wish. You will receive a copy of the transcript to approve, adjust, or update as needed before it is used in the final thesis.

To assist you in reaching a decision, I am attaching the following letters:

- A copy of the consent document, as approved by the University of Alberta Research Ethics Board.
- b) A copy of the approval for the study provided by the Research Ethics Board.

Please reach out with any questions. If you know of anyone who you think would be a good fit for this study, please forward them this email and they can reach out for more information at <u>bbertram@ualberta.ca</u>

I look forward to hearing from you! Brianne Bertram she/her bbertram@ualberta.ca Principal Investigator: Brianne Bertram, Graduate student, bbertram@ualberta.ca

Supervisor: Sandra Garvie-Lok, Professor, sandra.garvie-lok@ualberta.ca

QUALITATIVE RESEARCH - INTERVIEW GUIDE DRAFT 3

INTRODUCTION

- Greetings
- Informed Consent
 - The informed consent form will be sent as a pdf ahead of time, this is to ensure there are no further questions or mind changes. At this stage, I will also confirm their anonymity status.
- Who are you? And positionality
 - Ask preferred pronouns for the meeting and future correspondence.
 - Job position etc.

INTERVIEW GUIDE

- Background
 - a. You were chosen for this interview because of your public interest in transgender status and forensic anthropologists. Why do you believe this research is important?
 - i. High-risk community
 - ii. Social stigma
 - b. Can you identify any physical skeletal markers, either established or potential, that you think could be used to assist in the identification of a transgender individual?
- 2. Existing literature and my own review
 - Tallman 2021 reported that 97.3% of the forensic specialists they surveyed agreed that research on this topic in forensic anthropology should work WITH transgender communities.
 - i. Do you think this is beneficial? Why?
 - ii. Which communities do you think we should be working with? Local transgender alliances? Or larger scale groups?
 - insgender annances? Of larger scale gro
 - Trans doe Task force....
 - b. This is not the first-time forensic anthropologists have been faced with calls for radical change in how we operate and perform identifications... I'm referring to the use of the term race and the shift to ancestry.
 - i. Forensic anthropologists have widely accepted the difference between ancestry and race, with race being a cultural construct. Do you think we will see a similar shift in how we report sex identification?
 - c. One argument against the use of puberty blockers for transgender children is that they cause a loss of bone mass density. Are you familiar with this idea?
 - Klink et al. 2015 studied 34 individuals aged 22 who had transitioned by using puberty blockers in adolescence. They found decreased areal BMD in the femoral neck.
 - They also found that cross-sex hormone therapy resulted in increased areal BMD scores but it was still not as high as those who had not undergone any treatments. However, this study did not take into account lifestyle and diet

and found varied results with an overall small increase in BMD from the control.

- iii. Hormonal Birth control are you familiar with the idea that transgender women will use oral contraception to assist in their transition even without doctor supervision? Are you familiar with oral contraception's BMD effects?
 - Almsetdt, Shoepe and Snow 2005 performed a DXA study on 18-25-year-old women and found a lower BMD than controls.
 - Alternatively, Garnero et al. 1995 found that in women over 39.5, those experiencing perimenopause, oral contraception dramatically decreased BMD loss.
 - DeCherney 1996 repeated the study on women over 35 and found virtually no effect on BMD from the use of oral contraception. He concluded that while it may slow down BMD loss, it does not lead to an increase.
 - a. What do you think this means for individuals who are transitioning? Do you think this will combine with other ideas about BMD loss and transitioning?
 - b. The majority of this research is older, from the late 90s or early 2000s. Do you think it is sufficient? What do you think we can do to find a more conclusive answer to this question regarding hormonal effects on BMD?
- c. Isa and Michael 2022 argue for a shift to ASAB–assigned sex at birth This would be a minor terminology change that removes one of the structural barriers for transgender identification.
 - If something like ASAB was included in official reports, how would you then indicate that the person's legal sex may vary from the ASAB?
 - ii. Do you think there would be backlash? Either from academic communities, the police who read your reports, or the broader community?
- d. Flaherty et al. 2023 used a CT scan of a transgender woman who had undergone 18 gender-affirming surgeries.
 - *Show image* If you were presented with this skull with no other information, what would your report say?
 - ii. They then ran her, measurements through FORDISC to see what the individual was classified as. FORDISC still determined the individual to be male, what do you think this means?
 - iii. In my thesis, I am proposing that forensic anthropologists can become familiar with signs of surgical scaring that can be indicative of gender affirming surgeries
 - How would you propose we report that information? → as pathologies? This is consistent with...
 - 2. What would you need to be able to say consistent with?
 - 3. What do you think is the future of this research? Thus far CT seems to be the only option, any other ideas?
- 3. Cultural Context
 - a. How often, if ever, do you receive additional information about a scene?

- ... or would you prefer to make your report based purely on skeletal assessment?
- Reporting
 - a. When presented with the signs discussed throughout this interview, what do you feel you need to feel confident in your analysis of transgender remains?
 - i. What about the difference between probable and possible? What word choice would you use and why?
 - ii. What could change or increase this confidence?
 - More published studies?
 - 2. More inclusive wording?
- 5. Finishing up
 - a. Is there anything else you would like to add?
 - b. What do you think the next steps of this research should be?
 - c. Is there anything you would have asked me if you were the interviewer?
 - d. Thank you and inform the participant they will have a copy of the transcript within the next month. After receiving the transcript copy, they will have two weeks to ask for any alterations, omissions, or additions, as well as a change to their anonymous status if desired.



FACULTY OF ARTS DEPARTMENT OF ANTHROPOLOGY

PARTICIPANT CONSENT FORM

Title of Study: Beyond Binary Identification: New ideas of gender and how they fit into forensic anthropology

Contact Information: Principal Investigator and Research/Study Coordinator: Brianne Bertram Name & Affiliation: Brianne Bertram, Department of Anthropology, University of Alberta Mailing Address: 13-5 Tory Building, Edmonton, Alberta, Canada, T6G2H4 Phone: 780-492-3879 Email: bbertram@ualberta.ca

Supervisor: Name & Affiliation: Sandra Garvie-Lok, Supervisor Mailing Address: 13-5 Tory Building, Edmonton, Alberta, Canada, T6G2H4 Phone: 780-492-3879 Email: sandra.garvie-lok@ualberta.ca

You are invited to take part in a research study. Before you take part, a member of the study team is available to explain the project and you are free to ask any questions about anything you do not understand. You will be given a copy of this form for your records.

Why am I being asked to take part in this research study?

You are being asked to participate in this interview because of your position, title, and experience as a forensic anthropologist. You are being approached specifically because your record of publications and presentations indicates expertise and interest in gender issues in forensic anthropology. The goal of this interview is to discuss how you would report on the appearance of skeletal remains of transgender individuals and the nature of the evidence needed to report on potential transgender status in a forensic case. You will be presented with some classes of evidence that can potentially be used in this identification and how you would interpret and report them in an official police report.

What is the reason for doing the study?

The purpose of this study is to gather information on forensic anthropologists' opinions on the issue of unidentified remains and unsolved missing persons cases involving transgender individuals. The study builds on the limited existing work on this topic, which has included a survey of forensic anthropologists' opinions. This study's goal is to expand this information with detailed interview data, focusing on how forensic anthropologists would report evidence of transgender status or lack thereof, why they would use certain terminology, and what is needed to increase their confidence in these answers. The results of this interview will contribute toward the principal investigator's MA thesis in biological anthropology, with findings presented and published after completion of the research.

What will I be asked to do?

You will be asked to participate in a conversation with the principal investigator where you will be asked your thoughts and opinions on topics including potential indicators of transgender status in a forensic case, how these would be presented in a police report and what further research is needed to increase practitioners' confidence in this work. This interview will take place over the course of about an hour, either in person or over Zoom depending on your location. If the interview happens on Zoom, you have the right to choose whether your camera is on or off. The interview will be recorded and then transcribed by the principal investigator, who will explore your answers and those of other participants for themes important to our understanding of forensic work on potential transgender cases. Before this happens, you will have the opportunity to review and approve the interview transcript.

What are the risks and discomforts? The principal investigator recognizes that topics revolving around transgender issues can be sensitive and may attract public attention. You will have the option to remain nameless or to be mentioned by name in the MA thesis and resulting presentations and publications. Because this is a sensitive and potentially controversial topic, participants who choose to include their names must be prepared that they may be challenged on their opinions in the academic sector or in social media.

It is not possible to know all of the risks that may happen in a study when presented to the public sphere, but we have taken all reasonable safeguards to minimize any known risks to a study participant. If we find out anything new during the course of this research which may change your willingness to be in the study, we will tell you about these findings.

What are the benefits to me?

The main benefit of this study will be the advancement of a new topic in the field of forensic anthropology that is becoming more relevant every day. This research will work towards solving the gap in knowledge about the skeletal remains of transgender individuals and address the growing concerns of missing person identification for an at-risk community. As an active practitioner in forensic anthropology, you may benefit indirectly from this advancement. However, there may not be any direct benefit to you.

Do I have to take part in the study? Being in this study is your choice. If you decide to participate in the study, you are still free to change your mind and withdraw your participation <u>until two weeks after receiving a copy of the interview transcript</u>. After that point, we cannot remove you from the study because data from your interview will be incorporated into the ongoing analysis. To withdraw from the study before this deadline, please contact Brianne Bertram at <u>bbertram@ualberta.ca</u>. Similarly, if you wish to change your choice on the inclusion of your name in this study, this can be done any time up to two weeks after receiving the interview transcript.

During your interview, you may refuse to answer any of the individual questions as you wish. After the interview you may choose to withdraw or alter individual responses by contacting Brianne Bertram at <u>bbertram@ualberta.ca</u>. This will be possible up to two weeks after you receive a copy of your interview transcript.

Will my information be kept private?

During this study we will do everything we can to make sure that all information you provide is kept private. No information relating to this study that includes your name will be released outside of the principal investigator's office or published unless you give us your express permission. Sometimes, by law, we may have to release your information with your name so we cannot guarantee absolute privacy. However, we will make every legal effort to make sure that your information is kept private

You will have the option to have your name included in the final version of this study. Participants may also choose not to have their names included and pseudonyms will be used (e.g. 'Participant A'). The only information that will be made known about these participants is that they are practicing forensic anthropologists. If you would like to use your real name, please indicate this on the signed consent form on the last page of this document.

For participants interviewed over Zoom, the call will be recorded to allow the primary investigator to make a transcript. This recording will be encrypted and stored on the primary investigator's secure Google Drive, not on the Zoom cloud. After the transcript is complete and approved by the participant, this video will be deleted and participants will be anonymized as they wish. The primary investigator will keep a document that includes the raw data from the study, including a document outlining which transcript is which in case of the need for follow-up questions.

During research studies, it is important that the data we get are accurate. For this reason, your data, including your name, may be looked at by people from the Research Ethics Board.

After the study is done, we will still need to securely store the data that was collected as part of the study. Your data will be stored on the principal investigator's secure Google Drive through the University of Alberta. It will also be stored on the research supervisor's secure Google Drive, as is the standard practice at the University of Alberta. At the University of Alberta, we keep data stored for a minimum of 5 years after the end of the study.

What if I have questions?

If you have any questions about the research now or later, please contact Brianne Bertram at bbertram@ualberta.ca

If you have any questions regarding your rights as a research participant, you may contact the University of Alberta Research Ethics Office at <u>reoffice@ualberta.ca</u> or 780-492-2615 and quote Ethics ID Pro00133307. This office is independent of the study investigators.

How do I indicate my agreement to be in this study?

By signing below, you understand:

- That you have read the above information and have had anything that you do not understand explained to you to your satisfaction.
- That you will be taking part in a research study.
- That you may freely leave the research study at any time.
- That you do not waive your legal rights by being in the study
- That the legal and professional obligations of the investigators and involved institutions are not changed by your taking part in this study.
SIGNATURE OF STUDY PARTICIPANT AND VERBAL CONSENT STATEMENT

Statement of Verbal Consent:

"I acknowledge that this research study has been explained to me. I have had the opportunity to ask questions and have them answered. I have been made aware that I can reach out with questions and who to contact. I agree to participate in the research study described in this document."

Name of Participant

Signature of Participant

Date

Do you wish to include your real name in the final versions of this study (including the accompanying master's thesis and any following publications? *Please indicate yes or no, if left unanswered, you will remain anonymous.*

Yes, I would like to use my real name.

No, I prefer not like to use my real name.

No, I would not like to use my real name but I would like to be thanked by name in the thesis acknowledgements.

SIGNATURE OF PERSON OBTAINING CONSENT

Name of Person Obtaining Consent

Contact Number

A copy of this consent form has been given to you to keep for your records and reference.